Utilization of By-products for Concrete: Development of One-part Mixing of Fly Ash, GGBFS and Silica Fume AAM and Applicability of Paper Sludge Ash

March 2022

MOHAMMAD IDRIS RASULI

199D9224

Architectural and Civil Engineering

Supervisor: Prof. Mitsuhiro SHIGEISHI

GRADUATE SCHOOL OF SCIENCE AND TECHNOLOGY KUMAMOTO UNIVERSITY

Abstract

OPC (ordinary Portland cement) is not an environmentally friendly construction material and it is a major contributor to global warming. AAM (alkali-activated material) which is mainly produced by aluminosilicate precursors and AA (alkaline activators) can be a substitute for the OPC due to its extraordinary properties such as high strength and durability. Moreover, AAM contains some by-products such as fly ash and slag, and compare to the OPC lesser amount of CO2 is emitted for their production. Therefore, they are adopted as environmentally friendly materials. Corrosive and viscosity effects of alkaline materials cause difficulty in handling the materials by common people and make it tough to cast them on site. Therefore, they are made as precast elements which increases the cost of their production. To overcome the problem, a one-part mixing method is proposed by some researchers but they found that the quality of AAM made by the one-part mixing method is lower than the one made by the two-part mixing method. Furthermore, low-calcium fly ash-based AAM has some problems such as a long setting time and low compressive strength when cured at ambient temperature curing. This research aimed to solve the aforementioned problems and develop a one-part mixing method of AAM.

The properties of one-part mixed AAM mortar were characterized. It was found that the quality of the AAM made by the aforementioned method was much lower than the conventional and OPC mortar. The compressive strength of the AAM was increased with an increase in slag amount but the flowability of the AAM was much lower due to the quick setting of the fresh mortar. The higher reactivity of the slag caused the problem. Therefore, sucrose was added to the mortar to delay the setting time. Sucrose increased the setting time and improved the flowability, compressive strength, and drying shrinkage of the AAM.

The flow and compressive strength properties of AAM made by one-part and two-part mixing methods were tested and compared. It was found that the AAM made by the two-part mixing method had much better flowability and compressive strength than the one-part mixed AAM. Therefore, the solubility test of solid alkaline activators was conducted and it was revealed that sodium metasilicate couldn't be dissolved well in a solvent at a temperature around 25°C. So, it was concluded that when 32% sodium metasilicate is used for making AAM using one-part mixing method, water or a solvent need to have a temperature of around 40°C. In addition, the properties of AAM made by sodium metasilicate were investigated. The AA solution made by sodium metasilicate was smooth and fluent but it was susceptible to a lower temperature than 30°C as a crystallization of the solution could easily occur. Hot mixing conditions played a positive role in increasing the flowability of the mortars. Compressive strength of 40MPa and flexural strength of 6.1 MPa were achieved by AAM using granular sodium metasilicate.

As mentioned before that low-calcium fly ash-based AAM has a much longer initial and final setting time and low compressive strength when cured at ambient temperature conditions. So, in this experiment, the influence of the type and concentration of sodium silicate was studied on the

setting time, flowability, and compressive strength of low-calcium fly ash-based AAM. It was found that the decreased concentration of sodium disilicate had a considerable positive effect on the setting time and compressive strength of AAM cured at ambient temperature.

In addition, this research also aimed to study the influence of recycled waste materials such as recycled coarse aggregate produced by pulsed power technology, paper sludge ash and silica fume on properties of ordinary Portland cement concrete. It was found that both silica fume and paper sludge ash showed their pozzolanic reactivity and improved some of the properties such as compressive strength of both conventional and recycled concrete. furthermore, high-quality recycled aggregate was produced by pulsed power technology and using the aggregate didn't have a negative effect on the properties of concrete.

Keywords: Alkali-activated material, One-part mixing, Alkaline activators, Fly ash, Slag, Solubility, Recycled Aggregate, Paper sludge ash.

Acknowledgements

First of all, I would like to thank my almighty Allah who has given me health and power to accomplish my thesis and doctoral program.

I would like to express my sincere gratitude to my supervisor Prof. Mr. Mitsuhiro Shigeishi for his outstanding support, guidance and instruction throughout the research and experiments. As he made me feel free to ask him questions during my study.

I would like to express my appreciation to the committee members: Prof. Mr. Kozo Onoue, Prof. Mr. Masahide Matsumura and Prof. Takuya Maruyama for their great support and valuable comments.

I sincerely appreciate Dr. Yuyun Tajunnisa for her valuable support and instruction throughout my doctoral course. Furthermore, I would like to thank Mr. Akifumi Yamamura for his contribution in my research. Furthermore, acknowledgments to Kumamoto university technicians, Yoshinori Toda, Ms. Arisa Hamasaki for their kind and valuable cooperation in conducting my experiments.

Moreover, special thanks to Kumamoto University and Japan government for financially supporting me throughout my academic journey in Japan.

At the end, I would like to extend my sincere thanks to my family, especially my wife for her kind cooperation and support all over the master and doctoral program. I pray for my late mother whom I lost during my study in Japan, she encouraged and backed me a lot. May Almighty Allah give her paradise.

Table of Contents

Abstracti	
Acknowledgementsiii	
Table of Contentsiv	
List of Tables xi	
List of Figures xiii	
1. Introduction1	
1.1 Background	
1.2 Objectives	
1.3 Thesis Outline	
References:	
2. A Review on Alkali-activated Materials	
2.1 History	
2.2 Reaction mechanism	
2.3 Geopolymer as a subset of alkali-activated materials	
2.4 Alkali-activated materials constituents 12	
2.4.1 Aluminosilicate precursors 12	
2.4.1.1 Fly ash 12	
2.4.1.2 Ground granulated blast-furnace slag	
2.4.1.3 Metakaolin	
2.4.2 Alkaline Activators	
2.4.2.1 Sodium silicate	
References:	

3. 	Characterization of the Alkali-activated materials Made by the One-part mixing Met	hod 20
	3.1 Introduction	20
	3.2 Objectives	21
	3.3 Experiment	21
	3.3.1 Materials	21
	a) Fly ash	21
	b) Ground granulated blast furnace slag	22
	c) Silica fume	22
	d) Ordinary portland cement (OPC)	22
	e) Standardized sand	22
	e) Sodium hydroxide (NaOH)	22
	d) Sodium silicate	22
	3.3.2 Mix proportion	23
	3.3.3 Mixing, casting and curing	24
	3.3.4 Flowability test	24
	3.3.5 Flexural and compressive strength test	24
	3.3.6 Drying shrinkage test	25
	3.4 Result	25
	3.4.1 Effect of aluminosilicate precursors mix proportion	25
	3.4.1.1 Flowability	25
	3.4.1.2 Strength	26
	3.4.1.3 Drying shrinkage	28
	3.4.2 Effect of AA ratio	29
	3.4.2.1 Flowability	29

3.4.2.2 Strength	
3.4.2.3 Drying shrinkage	
3.4.3 Effect NaOH molarity	
3.4.3.1 Flowability	
3.4.3.2 Strength	
3.4.3.3 Drying shrinkage	
3.4.4 Effect of the production method	
3.4.4.1 Flowability	
3.4.4.2 Strength	
3.4.4.3 Drying shrinkage	
3.5 Conclusion	
References	
4. Retardation Behaviour of Sucrose on Alkali-activated Materials	
4. Retardation Behaviour of Sucrose on Alkali-activated Materials	
 4. Retardation Behaviour of Sucrose on Alkali-activated Materials 4.1 Introduction 4.2 Objective 	
 4. Retardation Behaviour of Sucrose on Alkali-activated Materials	42 42 42 42 42
 4. Retardation Behaviour of Sucrose on Alkali-activated Materials 4.1 Introduction 4.2 Objective 4.3 Experiment 4.3.1 Materials 	42 42 42 42 42 42 42 42
 4. Retardation Behaviour of Sucrose on Alkali-activated Materials	42 42 42 42 42 42 42 42 42 43
 4. Retardation Behaviour of Sucrose on Alkali-activated Materials 4.1 Introduction 4.2 Objective 4.3 Experiment 4.3.1 Materials 4.3.2 Mortar preparation and experimental program 4.3.3 Paste preparation and setting time test 	42 42 42 42 42 42 42 42 42 43
 4. Retardation Behaviour of Sucrose on Alkali-activated Materials 4.1 Introduction 4.2 Objective 4.3 Experiment 4.3.1 Materials 4.3.2 Mortar preparation and experimental program 4.3.3 Paste preparation and setting time test 4.4 Result 	42 42 42 42 42 42 42 42 43 43 43 43
 4. Retardation Behaviour of Sucrose on Alkali-activated Materials	42 42 42 42 42 42 42 42 42 42 42 42 43 43 45
 4. Retardation Behaviour of Sucrose on Alkali-activated Materials 4.1 Introduction 4.2 Objective 4.3 Experiment 4.3.1 Materials 4.3.2 Mortar preparation and experimental program 4.3.3 Paste preparation and setting time test 4.4 Result 4.4.1 Setting time 4.4.2 Flowability 	42 42 42 42 42 42 42 42 42 42 42 42 43 43 45 45 46
 4. Retardation Behaviour of Sucrose on Alkali-activated Materials 4.1 Introduction 4.2 Objective 4.3 Experiment 4.3.1 Materials 4.3.2 Mortar preparation and experimental program 4.3.3 Paste preparation and setting time test 4.4 Result 4.4.1 Setting time 4.4.2 Flowability 4.4.3 Strength 	42 42 42 42 42 42 42 42 42 42 42 43 43 45 45 46 47

4.5 Conclusion	
Reference	
5. A Consideration on the One-part Mixing Method of Alkali-activate Comparison with the Two-part Mixing Method and Solubility Proble	ed Materials: Property em of Sodium Silicate 51
5.1 Introduction	
5.2 Objective	
5.3. Experiments	
5.3.1 Material	
5.3.2 Making alkali solution	
5.3.3 Comparison of the properties of samples made by the one-part method.	and two-part mixing 53
5.3.4 Experimental program for characterization of AAM made by s pentahydrate solution	odium metasilicate
5.3.4.1 Preparation of the aluminosilicate precursors	
5.3.4.2 Mixing, casting and curing	
5.4 Result and discussion	
5.4.1 Comparison of the properties of samples made by the one-part method.	and two-part mixing
5.4.1.1 Flowability	
5.4.1.2 Compressive strength	
5.4.2 Solubility of the solid alkaline materials	
5.4.3 Characterization of AAM made by the sodium metasilicate per	ntahydrate solution 58
5.4.3.1 Flowability	
5.4.3.2 Compressive and flexural strength	
5.4.3.3 Drying shrinkage	
5.5 Conclusion	

Reference:
6. A Study on The Influence of Sodium Silicate Concentration and Modulus Ratio, Temperature of the Raw Materials, Sucrose and Curing Conditions on the Properties of Low-Calcium Fly Ash-Based Alkali-activated Materials
6.1. Introduction
6.2. Objectives
6.3. Experiment
6.3.1 Materials 67
6.3.2 Mix proportion and samples detail
6.3.3 Paste preparation and setting time test70
6.3.4 Mortar preparation, flow and compressive strength test
6.4 Result 70
6.4.1 Influence of the sodium silicate concentration70
6.4.1.1 Setting time
6.4.1.2 Flowability
6.4.1.3 Compressive strength72
6.4.2 Influence of the SiO ₂ :Na ₂ O ratio72
6.4.2.1 Setting time
6.4.2.2 Flowability
6.4.2.3 Compressive strength73
6.4.3 Influence of using sucrose74
6.4.3.1 Setting time
6.4.3.2 Flowability and compressive strength75
6.4.4 Influence of the temperature of the raw materials
6.4.4.1 Setting time

6.4.4.2 Flowability	77
6.4.4.3 Compressive strength	
6.4.5 Effect of the curing condition on compressive strength	79
6.5 Conclusion	
References	
7. Fundamental Investigation of Recycled Concrete using Paper Sludge Ash Fume	and Silica
7.1. Introduction	
7.1.2 Significance of the research	
7.2 Objectives	
7.3 Experiment	
7.3.1 Materials	
7.3.1.1 Binders	
7.3.1.2 Aggregates	
7.3.2 Procedure and experiment	
7.3.2.1 Mix proportion	
7.3.2.2 Mixing	
7.3.2.3 Slump test, specimen casting, and curing	
7.3.2.4 Compression test	
7.3.2.5 Acoustic Emission (AE) behavior test	
7.4 Result and discussion	
7.4.1 Workability	
7.4.2 Compressive strength	
7.4.3 Acoustic emission (AE) behavior	
7.5 Conclusion	

8.	. Conclusion	, 96
	References	. 95
	Future research	, 94

List of Tables

Table 2.1 Chemical composition of high and low-calcium fly ash [49]	. 13
Table 2.2 Chemical composition of GGBFS [53]	. 13
Table 2.3 Molar ratio for commercial silicates [34]	. 14
Table 3.1 Specific gravity (S.G.), the chemical composition of materials in weight percentage (wt. %) by XRF, and chemical analysis of standardized sand	. 22
Table 3.2 Mix design of AAM and OPC mortars	. 23
Table 3.3 Mix proportion (kg/m ³)	. 23
Table 3.4 The mix proportion and flowability specification of the samples	. 26
Table 3.5 Mix proportion, AA ratio and flow value specification of the samples	. 29
Table 3.6 Mix proportion, NaOH molarity and flow value specification of the samples	. 32
Table 3.7 Mix proportion, production method and flow value specification of the samples	. 35
Table 4.1 Mix design of the mortar used in this experiment.	. 43
Table 4.2 Mix design and composition characteristics of pastes.	. 44
Table 5.1 The specification of each type of mortar	. 53
Table 5.2 Percentages of the aluminum silicate precursors in each type of the sample	. 54
Table 5.3 Mixture proportions of AAM mortars (kg/m ³)	. 54
Table 6.1 Specific gravity (S.G.), the chemical composition of materials in weight percentage (wt. %) by XRF, and chemical analysis of standardized sand	. 67
Table 6.2 Mix proportion and detail of each paste type	. 68
Table 6.3 Mix proportion and the details of each mortar type	. 69
Table 7.1 XRF of the cement, paper sludge ash, and silica fume	. 85
Table 7.2 Size distribution of fine aggregate	. 85
Table 7.3 Properties of coarse aggregate	. 86
Table 7.4 Size distribution of coarse aggregate	. 87

Table 7.5 Mix proportion (kg/m ³)
---	---

List of Figures

Figure 2.1 Reaction mechanism and final product of AAM consisting of high and low calcium- based aluminosilicate precursors [21]	0
Figure 2.2 AAM and its subsets according to Ca and Al contents [24] 1	1
Figure 3.1 Effect of aluminosilicate mix proportion on the flowability	:6
Figure 3.2 Effect of aluminosilicate mix proportion on the compressive strength	27
Figure 3.3 Effect of aluminosilicate mix proportion on the flexural strength	28
Figure 3.4 Effect of aluminosilicate mix proportion on the flexural strength	28
Figure 3.5 Effect of AA ratio on the flowability	0
Figure 3.6 Effect of AA ratio on the compressive strength	0
Figure 3.7 Effect of AA ratio on the flexural strength	1
Figure 3.8 Effect of AA ratio on the drying shrinkage	51
Figure 3.9 Effect of NaOH molarity on the flowability	52
Figure 3.10 Effect of NaOH molarity on the compressive strength	3
Figure 3.11 Effect of NaOH molarity on the flexural strength	3
Figure 3.12 Effect of NaOH molarity on the drying shrinkage	4
Figure 3.13 Effect of the production method on the flowability of AAM	5
Figure 3.14 Effect of the production method on the 28-day compressive strength of AAM 3	6
Figure 3.15 Effect of the production method on the flexural strength of AAM	6
Figure 3.16 Effect of the production method on the drying shrinkage of AAM	7
Figure 4.1 Setting time test: (A) Vicat apparatus, (B) Paste with marks of the needles	-5
Figure 4.2 Setting test result of the P1 and P1S	-5
Figure 4.3 Setting test result of the P2 and P2S	6
Figure 4.4 Flowability test result	7
Figure 4.5 Compressive strength result	7

Figure 4.6 Flexural strength result	48
Figure 4.7 Drying shrinkage result	49
Figure 5.1. Making alkaline solution: (A, B) Water Bath Shaker, (C) AA solution after dissolution process, (D) Crystallization of the solutes, (E) An incubator, (F) AA solution after heat treatment in the incubator.	r 53
Figure 5.2 Comparison of the flowability	56
Figure 5.3 Comparison of the compressive strength	57
Figure 5.4 Flow value of M4	59
Figure 5.5 Flow value (mm)	60
Figure 5.6 The compressive strength (MPa)	60
Figure 5.7 The flexural strength (MPa)	61
Figure 5.8 Shrinkage test result	62
Figure 6.1 Influence of the change in concentration of sodium silicate solution on the setting to of the pastes	time 71
Figure 6.2 Influence of the change in concentration of sodium silicate on the flowability of A	AM 71
Figure 6.3 Influence of the change in concentration of sodium silicate on the compressive strength of AAM	72
Figure 6.4 Influence of the SiO ₂ :Na ₂ O ratio on the setting time	73
Figure 6.5 Influence of the SiO ₂ :Na ₂ O ratio on the flowability	73
Figure 6.6 Influence of the SiO ₂ :Na ₂ O ratio on the compressive strength	74
Figure 6.7 Influence of sucrose on the setting time	75
Figure 6.8 Influence of sucrose on the flowability	75
Figure 6.9 Influence of sucrose on the compressive strength	76
Figure 6.10 Influence of the temperature of the raw materials on the setting time of AAM	77
Figure 6.11 Influence of the temperature of the raw materials on the flowability	78
Figure 6.12 Influence of the temperature of the raw materials on the compressive strength	79

Figure 6.13 Effect of the curing condition on the compressive strength	80
Figure 7.1 SEM of OPC and PSA	84
Figure 7.2 Recycled aggregate	86
Figure 7.3 Slump test and samples after casting in Molds	89
Figure 7.4 Parameters of acoustic emission waveform [10]	90
Figure 7.5 Slump value of the concrete samples	91
Figure 7.6 The 28-day compressive strength	91
Figure 7.7 The 56-day compressive strength	92
Figure 7.8 Relationship between AE hits and stress level for the concrete samples	93

Chapter 1

Introduction

1.1 Background

Portland cement (PC) concrete has some remarkable properties such as the ability to cast in different shapes and sizes, high compressive strength, and reasonable cost. Moreover, its reinforcement by steel makes it an extraordinary construction material. According to these facts, Portland cement concrete is the most commonly used construction material in the world. Annual global production of ordinary Portland cement (OPC) reaches 4 billion tons [1]. But according to the statistics OPC is responsible for the emission of 8% CO₂ of the globe [2]. Therefore, they are not environmentally friendly materials and one of the major causes of global warming which is a serious issue nowadays.

Moreover, waste materials such as concrete debris, waste paper and by-products of energy production power plants such as coal ash have a negative impact on the environment because the majority of them are disposed of in landfills. Therefore, recycling waste materials not only has environmental benefits but, in some cases, it has economic advantages as well.

When aluminum silicate materials such as fly ash and slag react with an alkali source, it produces a material that has binding properties (3,4). These kinds of materials are called Geopolymer or alkali-activated materials (AAM). Compared to OPC, AAM is a more environmentally friendly material because they emit a much lower amount of CO₂ to the atmosphere [5, 6]. Moreover, they contain some by-products as their primary constituents, such as fly ash and slags [7].

Besides being environmentally friendly materials, AAMs have some other advantages, such as obtaining high early strength and superior mechanical properties [8,9]. They have better resistance to fire, acid attack, and alkali-silica reaction [10, 11,12].

AAM and geopolymer are made conventionally by the reaction of aluminosilicate precursors such as fly ash and slag, and aqueous alkaline solutions such as hydroxides, silicate and carbonate. This method of making the materials is called the two-part mixing method [13, 14, 15]. Handling and working with corrosive and viscous alkaline solutions cause difficulty in its application, and there is a need for special skill and caution for working with it. This limits the material's production in factories for precast purpose (16-18).

One-part mixing method or dry mixing method is being studied by some researchers [16, 17, 18] in which aluminosilicate precursors and alkali materials are used in solid-state, and water is just added for their reaction purpose. Cast-in situ or cast-in-place, and secure application of the materials are the main aims of studying this method [16-18].

Some researchers reported [19, 20] that the AAM made by the one-part mixing method had lower mechanical strength than the one made by the two-part mixing method. Moreover, Ouyang et al [19] found some undissolved particles of sodium metasilicate pentahydrate by studying the microstructure of the samples. This may negatively influence the properties of the product. Basically, the alkaline solution is the main part of geopolymer and AAM for geopolymerization and activation of the aluminosilicate precursors to make the binder [21]. It is essential that the solid alkali activators dissolve well to obtain the aqueous solution during the mixing procedure. Hence, modification of the mixing method is required by studying the alkaline materials' solubility in different solvent temperatures.

One of the aims of this study is to identify the existing problem in the one-part mixing method, modify it to equalize the quality of the one-part mixed AAM to the one made by a two-part mixing method.

Low calcium fly ash-based AAM has some extraordinary advantages. They possess geopolymeric characteristics such as low shrinkage and high fire resistance [1]. But they have much higher setting times, and it is due to their slow reactivity. Therefore, they are cured at higher temperatures [22,23]. Incorporating the slag with fly ash improves the reactivity of the aluminosilicate precursors that result in hardening of the samples at ambient temperature and modify some other properties such as compressive strength, flexural strength, and density of concrete [24]. As the slag is very active and increases the reactivity and hydration of the materials [25], this causes the materials to emit a higher amount of heat during the reaction process than fly ash-based AAMs [26]. Therefore, the drying shrinkage of alkali-activated materials with slag is much higher than the other types [24, 27].

The retarding behavior of sucrose on ordinary Portland cement materials is reported by some researchers, and they also concluded that sucrose improves the microstructure of ordinary Portland cement concrete [28,29]. Therefore, the drying shrinkage of the slag-based alkali-activated materials may be amended by using sucrose. Moreover, to better understand the retarding influence of sucrose on alkali-activated materials, the setting time test needs to be conducted.

As mentioned before, low calcium fly ash-based geopolymer or the alkali-activated material has a longer setting time when cured at ambient temperature conditions, but they have relatively low drying shrinkage compared to the slag-based AAM [23]. The longer setting time, make it almost impossible to be used and cast at the site of the construction. The viscosity of the water glass and the rheological problem of traditional geopolymer and AAM [30,31,32] are other problems that have a negative influence on the applicability of the material. However, different methods need to be applied such as preheating of the materials, changing the type and concentration of the alkaline solution to overcome both the setting time and rheological problems. Therefore, this research also aims to solve the problem related to the low-calcium-based AAM and make the material to be cured at ambient temperature conditions and appropriate for use in the structural field.

As stated above, materials such as concrete debris and waste paper have a negative influence on the environment. Modification of this kind of materials and using them in construction can reduce their disposal in landfills. Therefore, in this study, the effect of high-quality recycled aggregate

produced by pulsed power technology and paper sludge ash as a pozzolan on some of the properties of OPC concrete will be investigated.

1.2 Objectives

- a) Properties characterization of AAM using one-part mixing method or dry mixing method.
- b) Evaluation of the retardation effect of sucrose on AAM.
- c) Comparison of the properties of AAM made by the one-part and two-part mixing method.
- d) Solubility test of sodium metasilicate pentahydrate (Na₂SiO₃.5H₂O) by a solvent with different temperatures.
- e) Characterization of an aqueous solution made by sodium hydroxide and sodium metasilicate pentahydrate.
- f) Evaluation of the effect of sodium silicate concentration and modulus, and the temperature of the mixing materials on properties of low-calcium fly ash-based AAM cured at ambient temperature.
- g) Study the influence of the paper sludge ash, silica fume and recycled aggregate on properties such as strength and AE behaviour of OPC concrete.

1.3 Thesis Outline

Chapter 1 presents the introduction section of this thesis, which is consists of the background, objectives, and outline of the thesis. The chapter illustrates environmental issues created by OPC concrete and the environmental and structural importance of AAM and geopolymer as an alternative for OPC. Moreover, problems associated with the application of the alkali-activated materials and the need for this study to solve the problem were discussed in this section.

Chapter 2 is an introduction to geopolymer and alkali-activated materials, and their constituent materials: aluminosilicate precursor and alkaline materials. This chapter will also explain the mechanism of geopolymerization, dissolution of the precursors by the alkaline materials, and consolidation process of the product.

Chapter 3 discusses the properties of the AAM made by the one-part mixing method. The influence of different ratios of aluminosilicate precursors such as fly ash, the slag and silica fume, the concentration of NaOH and Na₂SiO₃: NaOH ratio on the flowability, mechanical strength, and drying shrinkage properties of the materials were discussed here. Moreover, the effect of sucrose on the properties of the AAM especially drying shrinkage explained in this chapter here.

Chapter 4 presents the retardation behaviour of sucrose. Moreover, the admixture's influence of the flowability, compressive strength and drying shrinkage is discussed here.

Chapter 5 of the thesis explains the solubility problem of sodium silicate in the one-part mixing method of alkali-activated materials. The solubility test of sodium metasilicate pentahydrate (Na2SiO3.5H2O) was conducted to reveal the optimum temperature of a solvent for totally dissolving the sodium metasilicate. Moreover, characterization of AAM made by the Na2SiO3.5H2O and NaOH solution is discussed.

Chapter 6 describes the influence of sodium silicate modulus and concentration on setting time, flowability and compressive strength of low-calcium fly ash-based AAM cured ambient (around 20°C) and in an incubator at 45°C. In addition to this, the influence of the raw material temperature, and sucrose on the aforementioned properties of the material are also presented in this chapter.

Chapter 7 discusses the properties of OPC concrete using pulsed powered recycled aggregate, waste paper sludge ash and micro-silica. Those properties are flowability, mechanical strength and acoustic emission behaviour.

Chapter 8 summarizes all the chapters in this dissertation.

References:

[1] Garside, M. (2021). World and U.S. cement production 2010-2019. Statista. https://www.statista.com/statistics/219343/cement-production-worldwide

[2] Andrew, R.M. (2019). Global CO₂ emissions from cement production, 1928–2018. Earth System Science Data Discussions. 1-67. DOI: 10.5194/essd-2019-152.

[3] Glukhovsky, V.D., Pashkov, I.A., Yavorsky, G.A., (1957). New building material, in Russian, Bulletin of Technical Information, GlavKievStroy, Kiev.

[4] Davidovits, J., (1984). Synthetic mineral polymer compound of the silicoaluminates family and preparation process, U.S. Patent 4,472,199.

[5] Van Deventer, J.S.J., Provis, J.L., Duxson, P., Brice, D.G. (2010) Chemical research and climate change as drivers in the commercial adoption of alkali activated materials. Waste and Biomass Valorization, 1, 145-155. DOI: 10.1007/s12649-010-9015-9

[6] Duxson, P., Provis, J.L., Grant, Van Deventer, J. (2007). The Role of Inorganic Polymer Technology in the Development of 'Green Concrete'. Cement and Concrete Research, 37, 1590-1597. DOI: 10.1016/j.cemconres.2007.08.018.

[7] Mehta, A., Siddique, R. (2016). An overview of geopolymers derived from industrial byproducts. Construction and Building Materials. DOI: DOI: 10.1016/j.conbuildmat.2016.09.136

[8] Fernández-Jiménez, A., Palomo, J.G., Puertas, F. (1999). Alkali-activated slag mortars: Mechanical strength behaviour. Cement and Concrete Research, 29, 1313-1321. DOI: 10.1016/S0008-8846(99)00154-4

[9] Hardjito, D., Wallah, S., Sumajouw, D., Rangan, B. (2004). Factors influencing the compressive strength of fly ash-based geopolymer concrete. Civil Engineering Dimension, 6, 88-93. Available at: https://www.researchgate.net/publication/26844029_Factors_influencing_the_compressive_strength_of_fly_ash-based_geopolymer_concrete

[10] Davidovits, J. (1991). Geopolymers - Inorganic polymeric new materials. Journal of Thermal Analysis, 37, 1633-1656.

[11] Kupwade-Patil, K., Allouche, E.N. (2013). Impact of Alkali Silica Reaction on Fly Ash-Based Geopolymer Concrete. Journal of Materials in Civil Engineering, 25, 131-139.

[12] Rashidian-Dezfouli, H., Rangaraju, P. R. (2021). Study on the effect of selected parameters on the alkali-silica reaction of aggregate in ground glass fiber and fly ash-based geopolymer mortars. Construction and Building Materials. 271. DOI: 10.1016/j.conbuildmat.2020.121549

[13]. Provis, J.L., Duxson, P., Kavalerova, E., Krivenko, P.V., Pan, Z., Puertas, F., van Deventer, J.S.J. Historical aspects and overview, In J.L. Provis, J.S.J. van Deventer (Eds.) (2014) Alkali-Activated Materials: State-of-the-Art Report, Rilem, T.C. 224-AAM, springer, Dordrecht, 11-57. DOI: 10.1007/978-94-007-7672-2_2

[14] Duxson, P., Fernández-Jiménez, A., Provis, J.L., Lukey, G.C., Palomo, A., Van Deventer, J.S.J. (2007) 'Geopolymer technology: The current state of the art. Journal of Materials Science, 42, 2917-2933. Available at: <u>https://link.springer.com/article/10.1007/s10853-006-0637-z</u>

[15] Luukkonen, T., Abdollahnejad, Z., Yliniemi, J., Kinnunen, P., Illikainen, M. (2018). One-part alkali-activated materials: A review. Cement and Concrete Research, 103, 21-34. DOI: 10.1016/j.cemconres.2017.10.001

[16] Nematollahi, B., Sanjayan, J., Qiu, J., Yang, E.-H. (2017). Micromechanics-based investigation of a sustainable ambient temperature cured one-part strain hardening geopolymer composite. Construction and Building Materials, 131, 552-563. DOI: 10.1016/j.conbuildmat.2016.11.117

[17]. Suwan, T., Fan, M. (2017). Effect of manufacturing process on the mechanisms and mechanical properties of fly ash-based geopolymer in ambient curing temperature. Materials and Manufacturing Processes, 32, 461-467. DOI: 10.1080/10426914.2016.1198013

[18] Hajimohammadi, A., van Deventer, J.S.J. (2017). Characterisation of one-part geopolymer binders made from fly ash. Waste and Biomass Valorization, 8, 225-233. DOI: 10.1007/s12649-016-9582-5

[19] Ouyang, S., Chen, W., Zhang, Z., Li, X., Zhu, W. (2020). Experimental study of one-part geopolymer using different alkali sources', Journal of Physics: Conference Series, 1605, 012155. DOI: 10.1088/1742-6596/1605/1/012155

[20] Nematollahi, B., Sanjayan, J., Shaikh, F.U.A. (2015). Synthesis of heat and ambient cured one-part geopolymer mixes with different grades of sodium silicate. Ceramics International, 41, 5696-5704. DOI: 10.1016/j.ceramint.2014.12.154

[21] Davidovits, J. (2011). Geopolymer Chemistry and Applications' 3rd ed., SaintQuentin. Available at: https://www.researchgate.net/publication/265076752_Geopolymer_Chemistry_and_Applications. Institut Géopolymère

[22] Ryu, G.S., Lee, Y.B., Koh, K.T., Chung, Y.S. (2013). The mechanical properties of fly ashbased geopolymer concrete with alkaline activators. Construction and Building Materials, 47, 409-418. DOI: 10.1016/j.conbuildmat.2013.05.069

[23] Tajunnisa, Y., Sugimoto, M., Sato, T., Shigeishi, M. (2017) 'A study on factors affecting geopolymerization of low calcium fly ash', International Journal of GEOMATE, 13, 100-107. DOI: 10.21660/2017.36.84153

[24] Tajunnisa, Y., Sugimoto, M., Uchinuno, T., Sato, T., Toda, Y., Hamasaki, A., Yoshinaga, T., Shida, K., Shigeishi, M. (2017). Effect of GGBFS and micro-silica on mechanical properties, shrinkage and microstructure of alkali-activated fly ash mortar. International Journal of GEOMATE, 13, 87-94. DOI: 10.21660/2017.39.11341

[25] Gruskovnjak, A., Lothenbach, B., Holzer, L., Figi, R., Winnefeld, F. (2006). Hydration of alkali-activated slag: Comparison with ordinary Portland cement. Advances in Cement Research. 18. 119-128. DOI: 10.1680/adcr.2006.18.3.119

[26] Han, F., He, X., Zhang, Z., Liua, J. (2017). Hydration heat of slag or fly ash in the composite binder at different temperatures. Thermochimica Acta, 655, 202-210. DOI: 10.1016/j.tca.2017.07.002

[27] Chi, M., Chang, J., Huang, R. (2012). Strength and Drying Shrinkage of Alkali-Activated Slag Paste and Mortar. Advances in Civil Engineering, 2012. DOI:10.1155/2012/579732

[28] Juenger, M.C.G., Jennings, H.M. (2002). New insights into the effects of sugar on the hydration and microstructure of cement pastes. Cement and Concrete Research, 32, 393-399. DOI: 10.1016/S0008-8846(01)00689-5

[29] Ahmad, S., Lawan, A., Al-Osta, M. (2020). Effect of sugar dosage on setting time, microstructure and strength of type I and type V Portland cements, Case Studies in Construction Materials, 13. DOI: 10.1016/j.cscm. 2020.e00364

[30] Provis J.L. (2009). Activating solution chemistry for geopolymers. In J.L. Provis, J.S.J. van Deventer. (Eds.), Geopolymers: Structures, Processing, Properties and Industrial Applications, Woodhead, Cambridge, UK (2009), pp. 50-71. DOI: 10.1533/9781845696382.1.50

[31] Xie, J., Kayali, O. (2016). Effect of superplasticiser on workability enhancement of Class F and Class C fly ash-based geopolymers. Construction and Building Materials, 122, 36-42. DOI: 10.1016/j.conbuildmat.2016.06.067

[32] Criado, M., Palomo, A., Fernández-Jiménez, A., Banfill, P.F.G. (2009). Alkali activated fly ash: Effect of admixtures on paste rheology. Rheologica Acta, 48, 447-455. DOI: 10.1007/s00397-008-0345-5

Chapter 2

A Review on Alkali-activated Materials

2.1 History

Alkali-activated material (AAM) was first reported by H. Kühl. He explained that when aluminosilicate (blast furnace slag) reacts with an alkali source produces a material that has cementitious and binding properties [1]. After that, Purdon [2] published his work regarding the material in 1940. He used blast furnace slag (BFS) and sodium hydroxide (NaOH) as an alkali activator. According to him, the process was in two steps, first, the liberation of silica aluminum and calcium hydroxide happens followed by the formation of silica and alumina hydrates. Later in 1950, the material was commercialized in Belgium and several buildings were made by the AAM [3].

In the mid-1950, Glukhovsky [4] developed the AAM and used low-calcium or calcium-free aluminosilicate precursors. These precursors were activated by alkaline solutions. He named the binder "soil cement" and the concrete "soil silicate". The binder was developed at a time when there was a shortage of Portland cement and need for an alternative in the former Soviet Union. The soil-cement was used widely in construction fields such as infrastructure and commercial fields [5]. Moreover, the AAM was commercialized and used in Northern Europe and China [6,7]. Glukhovsky [10] also investigated the alkali-activated blast furnace slag and described the hydrated products as calcium silicate hydrates and calcium and sodium aluminosilicate, considering that when clay mineral is activated by an alkali, converts to aluminum silicate hydrate(zeolite). He [11] categorized the alkaline activator into the 6 groups, where M is an alkali ion:

- 1. Alkalis, MOH
- 2. Weak acid salt, M₂CO₃, M₃PO₄, MF
- 3. Silicate, M₂O
- 4. Aluminates, M₂O.nAl₂O₃.
- 5. Aluminosilicate, M₂O.Al₂O₃. (2-6) SiO₂.
- 6. Strong acid salts, M_2SO_4 .

Davidovits [8] from France, developed the metakaolin AAM and started to commercialize the material as a fire-resistant product. His effort led to the commercialization of a high-early strength hybrid Portland/ alkali-activated concrete named Pyrament in North America [8]. Wastiels et al. were the first to show the activation of fly ash by the alkaline solution. A binder from the activation of fly ash and blast furnace slag (BFS) by an alkali was discovered and commercialized in the 1990s. This material was used as an acid resistance for pipe production [9]. Since then, studies on AAM have grown dramatically. This is to modify the material and to increase their applicability so as to be an alternative for Portland cement [12].

2.2 Reaction mechanism

The reaction mechanism of alkali-activated materials has not been fully understood till now. But researchers agree that the mechanism is much dependent on the prime materials such as aluminosilicate precursors and the activator solutions [13]. According to Glukhovsky [10], the mechanism is based on a destruction-condensation reaction. As first the aluminosilicate materials are destructed by the high PH of the alkaline solution, the destructed materials are transformed in a colloid phase, and then the destructed product is accumulated and interaction happens between them to form a condensed structure. So, the reaction mechanism occurs in three-stage of destruction, accumulation and condensation.

Other authors [14,15] also agree with the proposed stage of the reaction mechanism by Glukhovsky. As all stage occurs simultaneously, it makes it difficult to precisely study the mechanism [16]. Alonso and Palomo [17] studied the heat evolution of metakaolin-based alkali-activated materials in which NaOH was used as an alkaline material. They found several phase peaks. One related to the dissolution of metakaolin, followed by the induction stage that had low heat release, and the last peak belonged to the final structure formation of the AAM. Jaarsveld et al. [18] concluded that the formation of geopolymer follows the same reaction procedure of zeolite: dissolution of the prime materials (aluminosilicate precursors) to the hydroxide ions, the orientation of the separated ions, and condensation of the structure in the polymeric system.

The difference between the reaction mechanism of OPC binder and AAM is that OPC is dissolved in water with an initial PH of 0, however, the alkalinity of water is reached around 13 when the dissolution is proceeding [5]. But in the case of AAM, the aluminosilicate precursors cannot be dissolved in neutral water but need a solution with high alkalinity to dissolve the prime materials [8,11,16]. Soluble silicates are often added to the alkaline solution to get the final product with good strength and durability, high alkalinity of the solution prevents the silica reaction in the mixture thus the setting is delayed but when the pH of the solution decreases, coagulating and polymerization occur [20].

Garcia-Lodeiro et.al [21] characterized the reaction mechanism of low and high calcium-based AAMs. According to them, the reaction procedure for both types of alkali-activated materials consist of three stages. first, dissolution of Si-O-Si, Al-O-Al, and Al-O-Si. This stage is called the nucleation stage. Followed by the polycondensation or coagulation stage, in which coagulated structure between dissolved composition from the precursor form, and at the last stage, C-A-S-H (calcium aluminum silicate hydrate) is produced for a high calcium-based system. In which calcium decrease the setting time of the product and provide higher early strength. But in the case of the low calcium system, the inorganic hardened and 3-dimensional polymer structure is built by the development of the crystals in the final stage of the reaction process. The final product is N-A-S-H (sodium aluminum silicate hydrate) gel which has a higher setting time and often requires thermal curing to develop strength.

For the activation of the high calcium-based aluminosilicate precursors, a solution with moderate alkalinity is required to obtain a calcium silicate hydrates-like phase as a reaction product. But for the dissolving or activating low-calcium and calcium-free prime materials, moderated or high alkaline activating solution is required to produce a zeolite-like phase material [16].

John L. Provis and Susan A. Bernal [21] summarized the reaction process of low and high calciummodel;



K Annu. Rev. Mater. Res. 44:299–327

Figure 2.1 Reaction mechanism and final product of AAM consisting of high and low calciumbased aluminosilicate precursors [21]

2.3 Geopolymer as a subset of alkali-activated materials

Alkali activated material is widely categorized: as any binders produced by the reaction of an alkaline salt with a silicate material [22, 23]. The silicate materials can be calcium silicate or aluminosilicate precursors such as fly ash and blast furnace slag. The alkaline activators consist of alkali hydroxide, silicates carbonate, sulfates, aluminates, or any other materials that can accelerate the dissolution of the prime materials such as aluminosilicate precursors [24].

Inorganic polymers [25] belong to the alkali-activated materials group. They are formed through a disordered silicate network. For their formation compare to CaO and Na2O, a high amount of silica or alumina is required. BFS based alkali-activated material, combined fly ash and slag and aluminosilicate-based binders are all can make inorganic polymers. As alkaline activators, silicates, hydroxides or, carbonate can activate slag and make it an inorganic polymer. But sulfate can't be used for making the binder [24].

Geopolymers [8] belong to inorganic polymer, a subset of the alkali-activated material. Geopolymers are formed by aluminosilicate precursor and are highly coordinated and make probably a Q4 network structure (a network which contains 4 atoms of Al in each repeating subunit) [26]. The calcium content is usually low for making geopolymer that possesses a zeolitic network structure [27]. The alkaline activators in such kinds of AAM are hydroxides and silicates [28]. Low calcium-based fly ash and calcined clay are usually used as prime materials for making geopolymer binders.



van Deventer, J.S.J., Provis, J.L., Duxson, P. et.al [24] describe each type of the binders such as alkali-activated material and geopolymer in the following graph.

Increasing Al content

Figure 2.2 AAM and its subsets according to Ca and Al contents [24]

2.4 Alkali-activated materials constituents

Alkali-activated materials are mainly consisting of aluminosilicate precursors and alkaline activators.

2.4.1 Aluminosilicate precursors

Aluminosilicate precursors are the prime materials of AAM [10]. They are fly ash, GGBFS, metakaolin, rice husk ash, and other materials which are rich in silica and alumina. They are dissolved by alkaline activators so as to rearrange and become reactive to produce gel [21]. Since the invention of the AAM, many kinds of aluminosilicate precursors have been introduced to the field.

Fly ash, GGBFS, and metakaolin are commonly used as aluminosilicate precursors [29]. Moreover, silica sources, such as rice husk ash [30, 31], silica fume [32], and some others [33] are occasionally used with the precursors in the AAM.

2.4.1.1 Fly ash

Fly ash which is also called coal ash or flue ash is a by-product of coal-based thermal power plants. They are fine particles that exhaust together with the flue gas during the combustion of coal. They are acquired by means of electrostatic precipitators. The ashes which fall under the combustion chamber are called bottom ash. Compare to the fly ash, they are coarser and heavier. Fly ash that is utilized in AAM has a glassy, hollow, spherical shape, and are mainly consisted of SiO2, Al2O3, Fe2O3 and CaO [34]. The particle size of the fly ash ranges from 1 to 20 μ m. The majority of the fly ash is disposed of in landfills [35]. However, some countries such as the USA recycle about 43% of waste material [36]. In Portland cement, fly ash is not just used as a pozzolan but is also used to modify the rheology of the fresh mortar and concrete and prevent alkali-aggregate reaction [37, 38].

Commonly fly ash is divided into two types, low-calcium fly ash or class F and high-calcium fly ash or class C [34]. Chemical composition of each type is illustrated in Table 1. Class F type of fly ash is more used due to its availability and better performance such as improvement of workability and shrinkage [39]. The other type is less available and has unstable properties, therefore, it is not often used in cementitious material. One of the major drawbacks related to class F or low-calcium fly ash is its slow reactivity that causes a delay in the consolidation of the material. To enhance the reaction and intensify the hardening process, thermal curing or incorporation with another High-Ca component is recommended [40,41].

Fly ash contains a significant amount of silica (SiO2) and alumina (Al2O3), especially the F class. Both of the materials have an influence on the properties of fly ash-based AAM. De Silva et.al reported that SiO2 content in the fly increases the setting time of the paste while alumina Al2O3 decreases it [42]. However, the silica content in the fly ash tends to decrease the porosity and modify the microstructure of AAM and therefore, the strength of the material is increased [42].

Ferrite oxide is another component of fly ash. This material mainly specifies the color of the fly ash. Moreover, it also increases the specific gravity of the final product [43]. However, Jimenez and Palomo state that the exact behavior of iron oxide on AAM is not well understood [44].

Both MgO and SO3 are present in low amounts in AAM. They are allowed at a maximum range of 5% [45,46]. MgO has an influence on increasing the pH and accelerating the setting of the AAM [47].

LOI or loss of ignition is the residue of the unburnt coal in the fly ash. The increased amount of unburnt coal has a higher demand for water, therefore, the compressive strength and microstructure of the AAM is negatively influenced [48]. Because of this, ASTM recommends that the amount of the LOI not to be exceeded over 6% by the weight of the fly ash.

Compound	Class F (%)	Class C (%)
SiO ₂	47.2-54	18-24.8
Al ₂ O ₃	27.7-34.9	12.1-14.9
Fe ₂ O ₃	3.6-11.5	6.3-7.8
CaO	1.3-4.1	13.9-49
Free lime content	0.1	18-25
MgO	1.4-2.5	1.9-2.8
SO ₃	0.1-0.9	5.5-9.1
Na ₂ O	0.2-1.6	0.5-2
K ₂ O	0.7-5.7	1-3

Table 2.1 Chemical composition of high and low-calcium fly ash [49]

2.4.1.2 Ground granulated blast-furnace slag

Ground granulated blast-furnace slag or GGBFS is another by-product that can be used as an aluminosilicate precursor in alkali-activated material. This material is produced by the grinding of molten slag that is obtained from the steel-making factory. The term slag is also used for GGBFS in some countries [50]. GGBFS or slag is a cementitious material and possesses a high amount of Ca that helps in modifying the mechanical strength and durability of a product [10,51]. Moreover, GGBFS also contains a significant amount of silica and alumina in its content that makes it suitable for use as cementitious material [52]. chemical composition of slag is depending on the type of steel and the manufacturing process [53]

SiO ₂	CaO	Al ₂ O ₃	MgO	Fe ₂ O ₃	S	Cr ₂ O ₃	Na ₂ O +K ₂ O	MnO ₂	P ₂ O ₅	TiO ₂
27- 40 %	30- 50%	5- 33%	1- 2.1%	<1%	<3%	0.003- 0.007%	1-3%	<2%	0.02- 0.09%	<3%

The particle size of the GGBFS or slag play important role in its reactivity. Particle size below 2 μ m is considered to have good reactivity while the particles with a higher size than 20 μ m are slow in reaction [54]. Therefore, for the better performance of GGBFS in the AAM system, the specific surface of the material needs to be in the range of 400 m²/kg to 600 m2/kg [53].

Using GGBFS in alkali-activated materials has some prominent advantages such as high early strength and durability [55], high resistance against fire, and little environmental effect [56]. But there are some disadvantages associated with the use of the material including the formation of cracks, high shrinkage, and volumetric instability [57].

2.4.1.3 Metakaolin

Metakaolin is produced by the hydroxylation of kaolinite Si_2O_5 , Al (OH)4 which occurs at around 750°C. Metakaolin has a significant amount of silica and alumina in the range of around 44.4 to 73% and 14.5 to 47.43 respectively, the particle size of metakaolin varies from 1.20 to 38 µm and surface area of 2.16 to 22 m2/g [58]. Metakaolin is effective when used with low calcium-based AAM such as fly ash Class F that provides an increase in the polycondensation rate to make denser nano-and micro-structure, that cause improvement in the mechanical properties of the AAM [59].

2.4.2 Alkaline Activators

Alkaline activators are one of the main parts of alkali-activated materials. They are essential in dissolving aluminosilicate sources and produce a binder or cementitious material. PH of the alkaline activator plays important role in the dissolution process, and a PH value of higher than 11.5 is required for this purpose, for slag-based AAM, the value is around 13-13.6 [61]. Alkali hydroxide and silicate generated higher alkalinity while sulfate and carbonate produce moderate PH values [61]. Sodium hydroxide and sodium silicate are the two prominent activators that are more available and usually used in making AAM [60]. However, sodium hydroxide creates higher alkalinity than sodium silicate, but sodium silicate contributes in developing the mechanical strength because they provide additional silica to the (C-S-H) gel [16]

2.4.2.1 Sodium silicate

Sodium silicate is belonging to the family of soluble silicate [34]. Soluble silicates are formed by alkali metal oxide M2O, silica, and water. M2O can be Na2O, K2O, or Li2O. The general formula of the soluble silicate is xSiO2:M2O. zH2O, x is the degree of polymerization and z is the number of water molecules [34].

According to the molar ratio (MR) (SiO₂:Na₂O), sodium silicate falls into the following categories.

NO.	Name of the silicate	Formula	MR (SiO ₂ :Na ₂ O)
1	Sodium orthosilicate	Na ₄ SiO ₄	0.5
2	Sodium metasilicate	Na ₂ SiO ₃	1.0
3	Sodium disilicate	Na ₂ Si ₂ O ₅	2
4	Sodium polysilicate	Na ₂ O.3.3SiO ₂	3.3

Table 2.3 Molar ratio for commercial silicates [34]

References:

[1] Kühl, H. (1908). Slag cement and process making the same, US Patent 900,939.

[2] Purdon, A.O. (1940). The action of alkalis on blast-furnace slag. J. Soc. Chem. Ind. Trans. Commun. 59:191–202

[3] Buchwald, A., Vanooteghem, M., Gruyaert E., Hilbig H., De Belie N. (2014). Purdocement: application of alkali-activated slag cement in Belgium in the 1950s. Mater. Struct. In press; DOI:10.1617/s11527-013- 0200-8

[4] Glukhovsky, V.D. (1959). Gruntosilikaty [Soil Silicates]. Kiev: Gosstroyizdat. 154 pp.

[5] Shi, C., Krivenko P.V., Roy D.M. (2006). Alkali-Activated Cements and Concretes. Abingdon, UK: Taylor & Francis. 376 pp. DOI: 10.1201/9781482266900

[6] Hakkinen, T. (1987). Durability of alkali-activated slag concrete. Nordic Concrete Research, 6, 81–94.

[7] Wang, S.D. (1991). Review of recent research on alkali-activated concrete in China. Magazine of Concrete Research, 43, 29–35. DOI: 10.1680/macr.1991.43.154.29

[8] Davidovits, J. (1991). Geopolymers—inorganic polymeric new materials. Journal of Thermal Analysis and Calorimetry. 37, 1633–1656. DOI: 10.1007/BF01912193

[9] Blaakmeer, J. (1994). Diabind: an alkali-activated slag fly ash binder for acid-resistant concrete. Advanced Cement Based Materials, 1, 275–76. DOI:10.1016/1065-7355(94)90036-1

[10] Glukhovsky, V.D., Rostovskaja, G.S., Rumyna, G.V. (1980). High strength slag alkaline cements. In: Proceedings of the seventh international congress on the chemistry of cement. 3. 164–8.

[11] Glukhovsky, V.D. (1981). Slag-alkali concretes produced from fine-grained aggregate Vishcha Shkolay, Kiev.

[12] Waldmann, D., & Thapa, V.B. (2018). A short review on alkali-activated binders and geopolymer binders. Parts of books: Contribution to collective works. Available at: <u>https://orbilu.uni.lu/bitstream/10993/35284/1/A%20short%20review.pdf</u>

[13] Pacheco-Torgal, F., Castro-Gomes, J., Jalali, S. (2008). Alkali-activated binders: A review: Part 1. Historical background, terminology, reaction mechanisms and hydration products, Construction and Building Materials, 22, (7), 1305-1314.
DOI: 10.1016/j.conbuildmat.2007.10.015

[14] Davidovits J. (1988). Geopolymer chemistry and properties. In: Proceedings of 1988 geopolymer conference. 1; 1988. 25–48.

[15] Jaarsveld, J.G.S., Deventer, J.S.J., Lorenzen, L. (1998). Factors affecting the immobilization of metals in geopolymerised fly ash. Metallurgical and Materials Transactions, 29, 283–91.

[16] Palomo, A., Grutzek, M.W., Blanco M.T. (1999). Alkali-activated fly ashes. A cement for the future. Cement Concrete Research, 29(8), 1323–1329. DOI: 10.1016/S0008-8846(98)00243-9

[17] Alonso, S., Palomo, A. (2001). Calorimetric study of alkaline activation of calcium hydroxide–metakaolin solid mixtures. Cement Concrete Research. (31), 25–30. DOI: 10.1016/S0008-8846(00)00435-X

[18] Jaarsveld, J.G.S., Deventer, J.SJ., Lukey, G.C., (2002). The effect of composition and temperature on the properties of fly ash and kaolinite based geopolymers. Chemical Engineering Journal. 89, 63–73. DOI: 10.1016/S1385-8947(02)00025-6

[19] Neville, A.M. (1997). Properties of Concrete, Longman, Essex, UK.

[20] Iler, R.K., (1979). The Chemistry of Silica: Solubility, Polymerization, Colloid and Surface Properties and Biochemistry of Silica, Wiley, New York, USA.

Available at: https://www.wiley.com/enus/The+Chemistry+of+Silica%3A+Solubility%2C+Polymerization%2C+Colloid+and+Surface+ Properties+and+Biochemistry+of+Silica-p-9780471024040

[21] Garcia-Lodeiro, I., Palomo. A., Fernández-Jiménez, A. (2015). An overview of the chemistry of alkali activated cement-based binders. Woodhead Publishing Limited. DOI: doi.org/10.1533/9781782422884. 1.19

[21] Provis, J.L., Bernal, S.A. (2014). Geopolymers and Related Alkali-Activated Materials. Annual Review of Materials Research. 44 (1), 299-327. Doi:10.1146/annurev-matsci-070813-113515

[22] Buchwald, A., Kaps, C., Hohmann, M. (2003). Alkali-activated binders and pozzolan cement binders—complete binder reaction or two sides of the same story? In: Proceedings of the 11th International Conference on the Chemistry of Cement, 1238–1246.

[23] Shi, C., Krivenko, P.V., Roy, D.M. (2006). Alkali-Activated Cements and Concretes. Taylor & Francis, Abingdon, UK.

[24] van Deventer, J.S.J., Provis, J.L., Duxson, P. Brice, D.J. (2010). Chemical Research and Climate Change as Drivers in the Commercial Adoption of Alkali Activated Materials. Waste Biomass Valor 1, 145–155. DOI: 10.1007/s12649-010-9015-9.

[25]. Duxson, P., Provis, J.L., Lukey, G.C., van Deventer, J.S.J. (2007). The role of inorganic polymer technology in the development of 'Green concrete'. Cement and Concrete Research, 37(12), 1590–1597. DOI: 10.1016/j.cemconres.2007.08.018

[26] Rahier, H., Simons, W., van Mele, B., Biesemans, M. (1997). Low temperature synthesized aluminosilicate glasses. 3. Influence of the composition of the silicate solution on production, structure and properties. Journal of Material Science. 32(9), 2237–2247 DOI: 10.1007/BF00355128

[27] Provis, J.L., Lukey, G.C., van Deventer, J.S.J. (2005). Do geopolymers actually contain nanocrystalline zeolites? —a reexamination of existing results. Chemical Material. 17(12), 3075–3085. DOI: 10.1021/cm050230i

[28] Provis, J.L. (2009). Activating solution chemistry for geopolymers. In: Provis, J.L., van Deventer, J.S.J. (eds.) Geopolymers: Structures, Processing, Properties and Industrial Applications, pp. 50–71. Woodhead, Cambridge, UK.

[29] Luukkonen, T., Abdollahnejad, Z., Yliniemi, J., Kinnunen, P., Illikainen, M. (2018). Onepart alkali-activated materials: A review. Cement and Concrete Research, 103, 21-34. DOI: 10.1016/j.cemconres.2017.10.001 DOI: 10.1016/j.cemconres.2017.10.001

[30] Hajimohammadi, A., van Deventer, J.S.J. (2016). Solid reactant-based geopolymers from rice hull ash and sodium aluminate. Waste Biomass Valorization. 8, 2131–2140. DOI: 10.1007/s12649-016-9735-6

[31] Sturm, P., Gluth, G.J.G., Brouwers, H.J.H., Kühne, H.-. (2016). Synthesizing one-part geopolymers from rice husk ash. Construction and Building Materials, 124. 961-966. DOI: 10.1016/j.conbuildmat.2016.08.017

[32] Ye, N., Yang, J., Liang, S., Hu, Y., Hu, J. (2016). Synthesis and strength optimization of onepart geopolymer based on red mud. Construction and Building Materials. 111 (34), 317-325. DOI: 10.1016/j.conbuildmat.2016.02.099

[33] Sturm, P., Gluth, G.J.G., Simon, S., Brouwers, H.J.H., Kühne, H.-. (2016). The effect of heat treatment on the mechanical and structural properties of one-part geopolymer-zeolite composites. Thermochimica Acta, 635, 41-58. DOI: 10.1016/j.tca.2016.04.015

[34] Davidovits, J. (2011).Geopolymer Chemistry and Applications, Third ed., SaintQuentin:InstitutGéopolymère.Availablehttps://www.researchgate.net/publication/265076752_Geopolymer_Chemistry_and_Applications

[35] Wikipedia contributors. (2021, July 25). Fly ash. In Wikipedia, The Free Encyclopedia.Retrieved00:08,July29,2021,from https://en.wikipedia.org/w/index.php?title=Fly_ash&oldid=1035336135

[36] American Coal Ash Association. Available at: <u>http://www.acaa-usa.org</u>

[37] Mindess, S. and Young, J. F. (1981). Concrete, Prentice-Hall, Inc., New Jersey, 671.

[38] Alasali, M.M., Malhotra, V.M. (1991). Role of Concrete Incorporating High Volumes of Fly Ash in Controlling Expansion due to Alkali-Aggregate Reaction. ACI Materials Journal, 88 (2), 159-163.

[39] Provis, J.L., Van Deventer, J.S.J. (2009). Geopolymers: structures, processing, properties and industrial applications. Woodhead publishing. <u>https://doi.org/10.1533/9781845696382</u>.

[40] Shekhovtsova, J., Zhernovsky, I., Kovtun, M., Kozhukhova. N, Zhernovskaya, I., Kearsley, E. (2018). Estimation of fly ash reactivity for use in alkali-activated cements - a step towards sustainable building material and waste utilization. Journal of Cleaner Production, 178, 22–33. DOI: 10.1016/j.jclepro.2017.12.270

[41] Abdel-Gawwad, H.A., García, S.R.V., Hassan, H.S. (2018). Thermal activation of air cooled slag to create one-part alkali activated cement. Ceramic International 44(12):14935–14939. DOI: 10.1016/j.ceramint.2018.05.089

[42] De Silva, P., Sagoe-Crenstil, K., Sirivivatnanon, V. (2007). Kinetics of geopolymerization: Role of Al_2O_3 and SiO_2 , Cement and Concrete Research, 37 (4), 512-518, DOI: 10.1016/j.cemconres.2007.01.003

[43] Wesche, K. (2004). Fly ash in concrete: properties and performance. (2004). Available at: https://www.academia.edu/6064988/Fly_Ash_in_Concrete_Properties_and_Performance.

[44] Fernández-Jiménez, A., Palomo, A. (2003). Characterization of fly ashes. Potential reactivity as alkaline cements, Fuel, 82(18), 2259±2265. DOI: 10.1016/S0016-2361(03)00194-7

[45] Bureau of Indian Standard (BIS), IS 3812:2003 Pulverized Fuel Ash-Specification, (2003)

[46] ASTM C618-03, Standard specification for coal fly ash and raw or calcined natural pozzolana for use in concrete, 'West Conshohocken, PA, (2003).

[47] Antoni, S. Wibiatma Wijaya, D. Hardjito. (2016). Factors Affecting the Setting Time of Fly Ash - Based Geopolymer, Material Science Forum, 841, 90±97. DOI: 10.4028/www.scientific.net/MSF.841.90

[48] Skvarla, J., Sisol, M., Botula, J., Kolesárová, M., I. Krinická, I. (2011). The potential use of fly ash with a high content of unburned carbon in geopolymers. Acta Geodynamica et Geomaterialia., 8(2), 123±132.

[49] Jarrige, A., (1971). Les cendres volantes, proprieties- application industrielles, ed. Ed. Eyolles, Paris.

[50] Ilyushechkin, A.Y., Roberts, D.G., French, D., Harris, D.J. (2012). IGCC Solids Disposal and Utilisation, Final Report for ANLEC Project 5- 0710-0065," CSIRO, Australia.

[51] Tajunnisa, Y., Sugimoto, M., Uchinuno, T., Sato, T., Toda, Y., Hamasaki, A., Yoshinaga, T., Shida, K., Shigeishi, M. (2017). Effect of GGBFS and micro-silica on mechanical properties,

shrinkage and microstructure of alkali-activated fly ash mortar. International Journal of GEOMATE, 13, 87-94. DOI: 10.21660/2017.39.11341

[52] Pacheco-Torgal, F., Labrincha, J., Leonelli, C., Palomo, A., Chindaprasit, P. (2015). Handbook of alkali-activated cements, mortars and concretes. Available at: DOI: 10.1016/C2013-0-16511-7

[53] Pacheco-Torgal, F., Labrincha, J., Leonelli, C., Palomo, A., Chindaprasit, P. (2014). Handbook of alkali-activated cements, mortars and concretes. Elsevier; 2014. DOI: 10.1016/C2013-0-16511-7

[54] Wang, P.Z., Trettin, R., Rudert, V. (2005). Effect of fineness and particle size distribution of granulated blast-furnace slag on the hydraulic reactivity in cement systems. Advances in Cement Research.17(4), 161–7. DOI:10.1680/adcr.2005.17.4.161

[55] Collins. F., Sanjayan, J.G. (2000). Effect of pore size distribution on drying shrinkage of alkali-activated slag concrete. Cement and Concrete Research. 30(9):1401–1406. DOI: 10.1016/S0008-8846(00)00327-6

[56] Yang, K.H., Song, J.K., Song K. (2013). Assessment of CO₂ reduction of alkali-activated concrete. Journal of Clean Production. 39, 265–272. DOI: 10.1016/j.jclepro.2012.08.001

[57] Oderji, S.Y., Chen, B., Ahmad, M.R., Shah, S.F.A. (2019). Fresh and hardened properties of one-part fly ash-based geopolymer binders cured at room temperature: effect of slag and alkali activators. Journal of Clean Production. 225, 1–10. DOI: 10.1016/j.jclepro.2019.03.290

[58] Rakhimova, N.R., Rakhimov, R.Z. (2019). Reaction products, structure and properties of alkali-activated metakaolin cements incorporated with supplementary materials - a review. Journal of Materials Research and Technology. 8(1), 1522–1531. DOI: 10.1016/j.jmrt.2018.07.006

[59] Torres-Carrasco, M., Reinosa, J.J., de la Rubia, M.A., Reyes, E., Alonso Peralta, F., Fernández, J.F. (2019). Critical aspects in the handling of reactive silica in cementitious materials: effectiveness of rice husk ash vs nano-silica in mortar dosage. Construction and Building Materials. 223:360–367. 10.1016/j.conbuildmat.2019.07.023

[60] Phoo-ngernkham, T., Maegawa, A., Mishima, N., Hatanaka, S., Chindaprasirt, P. (2015). Effects of sodium hydroxide and sodium silicate solutions on compressive and shear bond strengths of FA–GBFS geopolymer. Construction and Building Materials. 91, 1–8.

Chapter 3. Characterization of the Alkali-activated materials Made by the One-part mixing Method

Chapter 3

Characterization of the Alkali-activated materials Made by the Onepart mixing Method

3.1 Introduction

Geopolymer and the Alkali-Activated Material (AAM) possess some excellent characteristics [1-4]. They are traditionally produced by a two-part mixing method, in which aluminosilicate precursors and alkaline aqueous solution are mixed [5-7]. But due to some drawbacks such as the corrosive effect of alkaline solution and workability problems related to the viscosity of alkaline materials especially water glass or sodium silicate solution, cause difficulty in the applicability of the material, and this cause making of the AAM to be just related to the expert people of this field [8].

Therefore, the one-part mixing method has been studied by some researchers to make ease in application of the AAM, in which aluminosilicate precursors and alkali materials are used in solid-state, and water is just added for their reaction purpose [8,9,10]. This is the same as making ordinary Portland cement materials. But some found the quality of the AAM made by the one-part mixing method to be lower than the one made by the traditional or conventional method [11,12].

The traditional AAM has been well characterized by researchers since its invention. The influence of many factors such as the mix proportion of aluminosilicate precursors, NaOH molarity, Na₂SiO₃: NaOH ratio and others on their properties are well studied. For the better characterization of the AAM made by the one-part mixing method, it's also necessary to study the effect of the aforementioned factors on the properties such as flowability and mechanical strength of the materials.

Aluminosilicate precursors are the major component of aluminosilicate materials. Their reaction with the alkaline materials results in making a cementitious material. Fly ash and GGBFS or slag are the two prominent aluminosilicate precursors [13]. Each of them has a significant influence on the properties of AAM. Low calcium fly ash improves some properties of AAM such as workability [14] and provides volumetric stability [15-17], but their setting needs a longer time [18]. Incorporation of slag with fly ash not only decrease the setting time [14] but can also improve the strength [14;19-20], and durability [21] of the material but shrinkage of the AAM is increased with the increment of slag in the mixture [22]. Therefore, an optimum mix proportion is needed to be considered to overcome the aforementioned problem. Besides fly ash and slag, silica fume is also used to modify the properties of the AAM [19].

Alkaline activators (AA) have a key role in making AAM as they react with the precursor and dissolve their compounds into smaller molecules. Sodium hydroxide or NaOH is one of the prominent and frequently used AA, its concentration or molarity influence has been studied on the
traditional AAM. Workability, mechanical strength and shrinkage are increased with the increment of the AA concentration [23], but some research shows that workability of the AAM is decreased with the higher molarity of AA [24].

Sodium silicate is used together with NaOH as an alkaline material. They provide extra SiO_2 to the AAM and improve some properties of the AAM. Therefore, the Na₂SiO₃: NaOH ratio has some significant influence on the properties of the AAM. A study [25] concluded that the compressive strength is improved with the increase in the ratio, they found the maximum compressive strength at the ratio of 2.5, but after that, the strength was declining. Nur Hamzah et al found the same result [26].

To study the effect of the production method and to compare the properties of AAM made by the one-part mixing method and the conventional AAM which is made by the two-part mixing method, the result of the flowability, the 28-day compressive and flexural strength, and drying shrinkage was taken from the research conducted by the Yuyun [27].

In this study, all the influence of all aforementioned factors will be evaluated on AAM made by the one-part mixing method.

3.2 Objectives

The objective of the chapter is to study the influence of the following factors on flowability, compressive and flexural strength, and shrinkage of the AAM made by the one-part mixing method;

- Aluminosilicate precursors mix proportion
- Molarity of NaOH
- Na₂SiO₃: NaOH
- Production method

3.3 Experiment

3.3.1 Materials

a) Fly ash

Two types of fly ash were used in this study, fly ash type F and PFBC type fly ash. Fly ash type F conforms to the JIS standard called fly ash A 6201. This type of fly ash which is also called low calcium fly ash was achieved from Kyushu Electric Power Co., Ltd. The other type which is called PFBC type fly ash was received from the Kyushu Electric Power Kanda Power Plant. The difference between both types of fly ash is in their composition and burning temperature. The PFBC fly ash burns at a lower temperature than the fly ash type F, and PFBC fly ash contain more CaO than the other type. Table 3.1 shows the composition and other specifications of both types of fly ash.

b) Ground granulated blast furnace slag

The specific surface area of the slag was 4000-4120 cm2 /g. The composition of the Ground Granulated Blast Furnace Slag (GGBS) or slag is shown in Table 3.2.

c) Silica fume

Silica fume conforming JIS A 6207 with a density of 2.22 g/cm³ was used in this study. The particle shape of the material was spherical.

d) Ordinary Portland Cement (OPC)

Ordinary Portland cement was not used in an alkali-activated material but was used for comparison purposes.

e) Standardized sand

Standardized sand conforming to the JIS R5201 standard was used as a fine aggregate.

Table 3.1 Specific gravity (S.G.), the chemical composition of materials in weight percentage (wt. %) by XRF, and chemical analysis of standardized sand.

Materials		Chemical composition (% weight)							
	SiO ₂	Al ₂ O ₃	Fe ₂ O ₃	CaO	Na ₂ O	H ₂ O	K ₂ O		
Fly ash type F	55.19	25.35	7.57	4.06				2.33	
PFBC	34.2							2.64	
Slag	33.8	13.4	0.7	41.7				2.91	
Micro-silica	93.67	0.83	1.3	0.31	0.4			2.22	
OPC	17.9	4.2	3.4	69	0.28			3.15	
Standardized sand	98.4	0.41	0.36	0.16	0.01	<0.2	0.01	2.64	

e) Sodium hydroxide (NaOH)

While making the conventional alkali-activated material, solid sodium hydroxide is dissolved in water to form an aqueous solution. But in this research, for the one-part mixing method, the solid NaOH was used together with the aluminosilicate precursors.

d) Sodium silicate

Granular sodium metasilicate pentahydrate (Na₂SiO₃.5H₂O) was used to make AAM using a onepart mixing method. Sodium oxide (Na₂O) and silicon dioxide (SiO₂ with amounts of 28 to 30% and 27 to 29% were used respectively. This means that Na₂O: SiO₂ modulus was around 1. For the AAM made by using a two-part mixing method, sodium silicate gel was used. Na₂O with 17 to 19% and SiO₂ 35 to 38% composed the material and the modulus was set to around 2.

3.3.2 Mix proportion

Mortar	Aluminosilicate mix	PFBC	Fly ash	GGBFS	Silica	OPC	NaOH	AA ratio	Extra	AA state
name	proportion	(%)	Type II	(%)	fume	(%)	(mol/L)	(g)	water	
			(%)		(%)				(%)	
GM1	PFBC100	100	0	0	0	0	16	1.0	10	Solid
GM2	FA100	0	100	0	0	0	16	1.0	10	Solid
GM3	FA80/GGBFS20	0	80	20	0	0	16	1.0	0	Solid
GM4	FA45/GGBFS45/SF10	0	45	45	10	0	16	1.0	10	Solid
GM5	FA42.5/GGBFS42.5/SF20	0	42.5	42.5	20	0	16	1.0	10	Solid
GM6	FA40/GGBFS40/SF20	0	40	40	20	0	16	1.0	10	Solid
AAM-16	FA42.5/GGBFS42.5/SF15	0	42.5	42.5	15	0	16	1.5	10	Solid
AAM-8	FA42.5/GGBFS42.5/SF15	0	42.5	42.5	15	0	8	1.5	10	Solid
AAM-8C	FA42.5/GGBFS42.5/SF15	0	42.5	42.5	15	0	8	1.5	0	Solution
OPC	OPC mortar	0	0	0	0	100	8		0	Solid

Table 3.2 Mix design of AAM and OPC mortars

Note: In the aluminosilicate mix proportion column, FA, GGBFS and SF stand for fly ash, slag and silica-fume or micro-silica respectively. The right-side value is the percentage of the material.

Table 3.3 Mix proportion (kg/m³)

Mortar	Sand	PFBC	Fly ash Type II	GGBFS	Micro- silica	OPC	NaOH	Na2SiO3.5H2 O	Water glass (Sodium disilicate)	Water	AE water reducing agent	L/B*
GM1	1536	511	0	0	0	0	61	76	0	190	0	0.6
GM2	1536	0	511	0	0	0	61	76	0	190	0	0.6
GM3	1536	0	417	104	0	0	62	77	0	193	0	0.6
GM4	1536	0	240	240	53	0	64	79	0	198	0	0.6
GM5	1536	0	226	226	80	0	64	79	0	197	0	0.6
GM6	1536	0	212	212	106	0	63	78	0	197	0	0.6
AAM-16	1536	0	229	229	81	0	52	96	0	197	0	0.6
AAM-8	1536	0	229	229	81	0	30	96	0	218	0	0.6
AAM-8C	1536	0	229	229	81	0	30	0	170	83	0	0.5
OPC	1536	0	0	0	0	512	0	0	0	256	10	0.5

*Liquid/Binder. L: AA solution, B: Aluminosilicate materials (Fly ash + GGBFS +Micro-silica).

3.3.3 Mixing, casting and curing

For each type of the mixture, four sets of 3 molds were prepared for the strength test (7, 14, 28 and 56 days), and one set of molds for the shrinkage test were prepared.

For the making of the mortar samples the following mixing procedure was applied;

First of all, the bowl of the mixer was cleaned with a dry cloth, in the case of one-part mixing method, aluminosilicate precursor and solid alkaline materials such as NaOH and Na₂SiO₃.5H₂O were mixed for one minute to make a paste, standardized sand was then added to the mixture and was mixed in low and high speeds, the duration for each was 30 seconds. Then, the mixing was stopped for 90 seconds followed by a high-speed mixing of 60 seconds. The same procedure of mixing was also used for making the ordinary Portland cement mortar.

For the two-part mixing method, the aluminosilicate precursors were first mixed for 2 minutes, then the alkaline solution and AE water reducing agent were added and mixed for two minutes to make a paste followed by the addition and mixing of standardized sand for 3 minutes. After the mixing procedure, the mortar was poured into the molds and vibrated for 2 minutes.

The surface of the molds was covered by plastic and put inside in a room with a temperature of around 20 °C. After 24 hours, the samples were demolded and each was wrapped and covered with plastic and cured at the aforementioned temperature until the day of the testing. But the samples which contained 100% fly ash couldn't be demolded due to their slow setting. Therefore, they were demolded at the age of 9 days.

OPC samples were demolded as same as AAM, but they were cured inside the water until the strength test day. But for the shrinkage test, they were cured for 7 days inside water and then cured at the same temperature that was used for the other type.

3.3.4 Flowability test

Flowability test was conducted according to JIS R 5201. The detail is as follow;

A flow table was used for measuring the flowability of the mortars. First of all, the flow table was put on a flat surface, and the steel plate and the cone were cleaned with a dry cloth. After the mixing procedure, the mortar was put in a flow cone which was properly fixed in the center position on the steel plate of the flow table. The mortar was put in the cone in two equal layers. Each layer was tamped 15 times by a tamped rod. Then the excessive mortar was cut off and the surface of the cone was flattened. Then the cone was slowly lifted upward. The plate was then risen and dropped 15 times. Next, the mortar was spread. The maximum direction and the opposed direction which was perpendicular to the maximum direction was measured. This procedure was done twice and the average value in millimeter (mm) was taken as a flow value of a mortar.

3.3.5 Flexural and compressive strength test

The flexural and compressive strength tests of the materials were carried out according to the Physical Testing Method for Cement, JIS R 5201. Each of the specimens was first tested for

flexural strength, then the two broken parts were taken for the compressive strength test, and the average value was taken as a value of the strength.

3.3.6 Drying shrinkage test

A dial gauge with an accuracy of 0.005 mm was used to test the drying shrinkage of samples. First of all, a reference rod with a specific length was measured. Then the specimen with attached pins with a length of 20 mm on both sides of its length was measured. This procedure was carried out twice for each sample. For the calculation of the drying shrinkage, first, the average length of the rod was subtracted from the average length of a sample to find the difference in their length. Deducting this length from the previous day's amount was the drying shrinkage value for each type of sample.

The measurement of the sample was started after 2 days to 7 days of demolding. GM2 which had 100% fly ash took a long time for setting, therefore, its measurement started after 9 days of the mixing.

For the drying shrinkage test, the specimen was kept uncovered at a temperature of around 20°C in a controlled temperature room. The samples were measured every day until they become 28 days old, then, the measurement was taken 2 to 3 times a week.

3.4 Result

3.4.1 Effect of aluminosilicate precursors mix proportion

3.4.1.1 Flowability

Figure 3.1 and Table 3.2 show the effect of the aluminosilicate mix proportion on the flowability of mortars.

GM2 which is composed of 100% fly ash had the highest flow value of 256 mm. GM7 and OPC which stand in the second and third positions have flow amounts of 163 mm and 149 mm respectively. The flowability of the other types is much lower.

From the flow result, it can be concluded that fly ash increased the flowability of mortars, and as the amount of fly ash is decreased and the amount of the slag is increased the flow value of the mortar is decreased. This is due to the spherical shape of the fly ash particles and their packing effect that causes low water retention. On the other hand, the angular particle shape of slag and also the presence of Ca cause low flowability of the material.

GM1, a PFBC fly ash mortar has much lower fluidity compared to the GM2. This is also due to the irregular particle shape of PFBC fly ash which results in high retention of water inside the particles.

Mix	The Mix proportion	Flow value (mm)
GM1	PFBC100	101
GM2	FA100	256
GM3	FA80G20	163
GM4	FA45G45M10	107
GM5	FA42.5G42.5M15	100
GM6	FA40G40M20	107
OPC	PC Mortar	149

Table 3.4 The	mix pro	portion an	d flowability	specification	of the	samples
	1	1	2	1		1





3.4.1.2 Strength

Figures 3.2 and 3.3 show the effect of aluminosilicate mix proportion on the compressive and flexural strength of the materials.

The compressive and flexural strength of both GM1 and GM2 are much lower in all ages. However, GM2 gained some strength at ages of 28 and 56 days which are 8.9 and 7.7 MPa for compressive strength and 1.8 and 1.5 MPa for flexural strength. However, the 56-day compressive and flexural strength of GM2 was lower than the 28-days', which is maybe attributed to the loss of mesopore water during curing at 18°C–20°C and 30%–50% relative humidity in the control temperature room, that loosens the mesopore walls. GM1's strength was not improved significantly at the later

ages. The slow reactivity of the fly ash causes both the GM1 and GM2 to have lower mechanical strength.

Both the compressive and flexural strength of the material was increased with the increment of the slag in the mixture. Replacement of 20% fly ash with slag significantly improved both the initial and final strength of the samples. Moreover, GM4, GM5 and GM6 had the almost same value of compressive and flexural strength for all ages which is significantly higher than the other type of AAM with the lower amount of GGBFS or slag. This is due to the presence of a higher amount of Ca element in the slag which contributes to the higher reactivity of the material and formation of C-A-S-H gel which reduces the porosity and condense the microstructure of the AAM [28].

Incorporation of the micro-silica in the mixture couldn't improve the strength of the AAM, in addition, the 56-day flexural strength was reduced with the increase of the material.

The compressive and flexural strength of all the alkali-activated materials made by the one-part mixing method was lower than the OPC mortar. This is in contradiction with the compressive result of the conventional AAM or the AAM made by a two-part mixing method [29]. Which indicates AAM's higher strength. A study shows that some even with the same mix proportions of GM4, GM5, and GM6 had the compressive strength higher than 65MPa which was higher than the OPC [27]. The decrease in strength in the one-part mixed method may be attributed to the deficiency in the reaction process between AA and the aluminosilicate precursors.

Moreover, the compressive strength result shows that the 56-day compressive strength of the OPC was decreased slightly from 62.1 MPa to 57.4 MPa. This is may be due to the enlargement of mesopores during the curing process. This defect is not usual in OPC concrete.



Figure 3.2 Effect of aluminosilicate mix proportion on the compressive strength



Figure 3.3 Effect of aluminosilicate mix proportion on the flexural strength



3.4.1.3 Drying shrinkage

Figure 3.4 Effect of aluminosilicate mix proportion on the flexural strength

The GM2 which has 100% fly ash had the lowest shrinkage value. The amount of shrinkage was less than 300 microstrain ($\mu\epsilon$). This is due to the low hydration rate of the fly ash mortar, moreover, the unreacted particles of fly ash can play a significant role in restraining the deformation of the material.

As far as GM1 is concerned, due to the increment of their length, the measurement couldn't be continued after the age of 7 days. This may be attributed to the accumulation of the high number of unreacted particles that cause enlargement in the size of the samples.

The shrinkage of the AAM is increased with an increase in the amount of slag. GM4 which had the highest amount of slag and micro-silica had the largest value of shrinkage that is around 3000 $\mu\epsilon$. This shrinkage value was more than twice of the OPC sample. GM5 and GM3 had the equal shrinkage value of 2000 $\mu\epsilon$. GM3 had a slightly lower drying shrinkage value compared to OPC samples. The high shrinkage of slag-based AAM is due to their high hydration rate that causes the retained water to evaporate quickly and result in capillary stress inside the samples.

3.4.2 Effect of AA ratio

3.4.2.1 Flowability

The mix proportion which was consisted of 42.5% fly ash, 42.5% slag, and 15% micro-silica was used to study the influence of the AA ratio, NaOH molarity, and extra water on the properties of the AAM.

Figure 3.5 shows the effect of the AA ratio on the flowability of the samples.

AAM-16 with a flow value of 111 mm had a slightly higher flowability compared to GM5's 100 mm. This indicates that the flowability of AAM made by the one-part mixing method is improved with the increase in AA ratio (Na₂SiO₃.H₂O: NaOH).

Mix	The mix proportion	Flowability	AA ratio (Na2SiO3.5H2O: NaOH)
GM5	FA42.5G42.5M15	100	1
AAM-16	FA42.5G42.5M15	111	1.5

Table 3.5 Mix proportion, AA ratio and flow value specification of the samples



Figure 3.5 Effect of AA ratio on the flowability

3.4.2.2 Strength

The compressive and flexural strength of the AAM-16 was higher than the GM4 at 7 days. At 14 days, both types of samples had an almost similar compressive strength of around 29 MPa. But for 28 and 56 days, the compressive and flexural strength of the sample with an AA ratio of 1.5 was lower than its counterpart. The result is in conflict with the previous finding [25,26] which indicates that the strength is improved with the increase in AA ratio. Compare to the GM5, the lower strength of AAM-16 may be attributed to the poor dissolution of the Na₂SiO₃.5H₂O particles which caused a lack in providing enough SiO₂ and PH to the matrix.



Figure 3.6 Effect of AA ratio on the compressive strength



Chapter 3. Characterization of the Alkali-activated materials Made by the One-part mixing Method

Figure 3.7 Effect of AA ratio on the flexural strength

3.4.2.3 Drying shrinkage

Figure 3.8 shows the effect of the AA ratio on drying shrinkage of the AAM made by the one-part mixing method. Both of the materials had higher drying shrinkage than OPC. The drying shrinkage was decreased with an increase in the AA ratio. This may be due to the presence of the unreacted sodium metasilicate particles which played a role to keep a stable skeleton.



Figure 3.8 Effect of AA ratio on the drying shrinkage

3.4.3 Effect NaOH molarity

3.4.3.1 Flowability

Figure 3.9 shows the influence of the NaOH solution molarity on the flow properties of the samples.

When the molarity increased from 8M to 16M, the flow of the mortar decreased by around 21%. This result confirms the finding of [24]. As higher molarity increased the viscosity and improved the reactivity of the constituent, the solidification of the mortar proceeded in quick time that reduced the flowability.

Table 3.6 Mix proportion, NaOH molarity and flow value specification of the samples

Mix	The mix proportion	Flowability	NaOH molarity
AAM-8	FA42.5G42.5M15	128	8M
AAM-16	FA42.5G42.5M15	111	16M



Figure 3.9 Effect of NaOH molarity on the flowability

3.4.3.2 Strength

The initial compressive and flexural strength of the mortar was increased with an increase in molarity of NaOH in the mixture. But the 28 and 56 days compressive and flexural strength of AAM-16 was lower than AAM-8. The maximum compressive strength of 41 MPa was achieved by the AAM-8 at 56 days. The higher strength of the mortar with lower NaOH concentration is

perhaps due to their higher flow rate which helps the aluminosilicate precursors and other constituents to disperse well through the mortar.



Figure 3.10 Effect of NaOH molarity on the compressive strength



Figure 3.11 Effect of NaOH molarity on the flexural strength

3.4.3.3 Drying shrinkage

Figure 3. 11 shows the effect of the NaOH molarity on the drying shrinkage of the AAM. There isn't any significant difference in the drying shrinkage amount between AAM-8 and AAM-16. But drying shrinkage of both types of mortar was much higher than the shrinkage of OPC. From the result, it can be perceived that the AA molarity doesn't have any influence on the drying shrinkage of AAM made by the one-part mixing method.



Figure 3.12 Effect of NaOH molarity on the drying shrinkage

3.4.4 Effect of the production method

This section compares AAM made by the one-part mixing method and conventional method. The conventional AAM was made with 54% sodium disilicate solution and a two-part mixing method was used for its production but the one-part AAM was made by 32% sodium metasilicate pentahydrate and the AA were mixed in a dry state.

The SiO₂:Na₂O ratio in sodium disilicate solution is 2, where this ratio is one for the sodium metasilicate. For comparing the compressive strength and flexural strength, a 28-day strength result was used. The 28-day strength result of the AAM-8C which is the conventional AAM was taken from the research conducted by Yuyun et.al [27]

3.4.4.1 Flowability

The flowability of AAM-8C was 181mm, and it is 41% more than the flowability of AAM made by the one-part (dry) mixing method. After the mixing of AAM-8, the hardening or consolidation of the mortar was quicker. This problem caused the fresh mortar to lose its workability.

Mix	The mix proportion	Flowability	Production method
AAM-8	FA42.5G42.5M15	128	One-part mixing (dry mixing)
AAM-8C	FA42.5G42.5M15	181	Conventional

Table 3.7 Mix proportion, production method and flow value specification of the samples



Figure 3.13 Effect of the production method on the flowability of AAM

3.4.4.2 Strength

Figures 3.13 and 3.14 show the influence of the AAM production method on the 28-day compressive and flexural strength of the AAM. AAM-8C which is a conventional type alkaliactivated material had much higher mechanical strength than the other type. The compressive and flexural strength of AAM-8C were 68 MPa and 39.8 MPa respectively. While AAM-8 had a compressive strength of 39.8 MPa and flexural strength of 6.3 MPa.

The higher compressive and flexural strength of the conventional type AAM was due to the higher SiO₂:Na₂O ratio of the water glass than the granular sodium metasilicate used in the AAM made by the one-part. Secondly, because of the low flowability and quick hardening of the AAM-8 mortar, the constituents of the mixture were not well dispersed in the mixture and cause lower strength of the material. This problem of a lower strength in the one-part mixing method is further analyzed in chapter 5 of this thesis.



Figure 3.14 Effect of the production method on the 28-day compressive strength of AAM



Figure 3.15 Effect of the production method on the flexural strength of AAM

3.4.4.3 Drying shrinkage

Figure 3.15 shows the effect of the production method of AAM on the drying shrinkage of the AAM.

The drying shrinkage of AAM-8C was significantly higher than AAM-8 until 60 days. But since then, the shrinkage of the AAM made by the conventional method was stopped and the length remained stable for the rest of the days. On the other hand, the drying shrinkage of AAM-8 was slower until around 40 days but the pace of the shrinkage was increased after that.

The higher shrinkage of AAM-8C at initial days was due to the higher hydration rate between the aluminosilicate precursors and AA solution. On the other hand, the low shrinkage value of the AAM-8 at the earlier time may be attributed to the shortage in dissolution rate of AA material which may result in a low reaction speed between the constituents of the material.



Figure 3.16 Effect of the production method on the drying shrinkage of AAM

3.5 Conclusion

AAM is conventionally produced by the two-part mixing method. The procedure of making is done by expert people. Handling the AA solution is difficult for common people as they are corrosive and cause danger to humans. Furthermore, the higher viscosity of AA solution used in making AAM by conventional method cause workability problem. With the aim to solve the aforementioned problems, a one-part mixing method was studied by some researchers. However, this kind of material has not been comprehensively characterized. Therefore, in this study, the effect of aluminosilicate precursors type and ratio, Na₂SiO₃: NaOH ratio, NaOH molarity, and the production method were studied. The following conclusion can be drawn from the result of this research;

- 1. Fly ash increases the flowability of AAM made by a one-part mixing method. This is due to the spherical shape and packing effect of its particles.
- 2. Overall, the compressive and flexural strength of all one-part mixed AAM made by different mix proportions of aluminosilicate precursors are lower than the OPC mortar.
- 3. GGBFS (slag) increases the early and final compressive and flexural strength of the AAM. Micro-silica doesn't have any significant influence on the strength of the materials.

- 4. PFBC and fly ash have lower initial and final strength. However, fly ash-based mortar slightly gain strength at the final ages, but still, that is much lower than the slag-based mortars.
- 5. 100% fly ash mortar has the lowest drying shrinkage among all samples and this is because of the low reactivity of their particles. But the drying shrinkage of the samples is increased with an increase in the amount of slag.
- 6. A higher hydration rate of slag causes the early loss of the mesopore water and creates capillary stress that results in drying shrinkage.
- 7. AA ratio doesn't have any significant influence on the flowability of the fresh mortars.
- 8. Both the compressive and flexural strength of AAM made by the one-part mixing method are decreased with the increase in AA ratio.
- 9. The drying shrinkage of AAM made by the one-part mixing method is increased with an increase in AA ratio.
- 10. The concentration of the NaOH has a slight influence on the flow and the strength properties of the materials. The flowability, compressive and flexural strength are improved with a decrease in the concentration of NaOH. But there isn't any effect on the drying shrinkage of the materials.
- 11. Conventional AAM has much higher flowability, compressive and flexural strength than the AAM made by the dry mixing method.
- 12. The drying shrinkage of the dry mixed AAM is lower in the early days but it is increased as the samples get older. This is in contradiction with the drying shrinkage result of conventional AAM, but after 100 days, the drying shrinkage amount of both types reach the almost same value.

References

[1] Van Deventer, J.S.J., Provis, J.L., Duxson, P., Brice, D.G. (2010). Chemical research and climate change as drivers in the commercial adoption of alkali activated materials. Waste and Biomass Valorization, 1, 145-155. DOI: 10.1007/s12649-010-9015-9

[2] Duxson, P., Provis, J.L., Lukey, G.C., Van Deventer, J.S.J. (2007). The role of inorganic polymer technology in the development of 'green concrete. Cement and Concrete Research, 37, 1590-1597. DOI: 10.1016/j.cemconres.2007.08.018

[3] Fernández-Jiménez, A., Palomo, J.G., Puertas, F. (1999). Alkali-activated slag mortars: Mechanical strength behaviour. Cement and Concrete Research, 29, 1313-1321. DOI: 10.1016/S0008-8846(99)00154-4

[4] Davidovits, J. (1991). Geopolymers - Inorganic polymeric new materials. Journal of Thermal Analysis. 37, 1633-1656. DOI: 10.1007/BF01912193

[5] Provis, J.L., Duxson, P., Kavalerova, E., Krivenko, P.V., Pan, Z., Puertas, F., van Deventer, J.S.J. Historical aspects and overview, In J.L. Provis, J.S.J. van Deventer (Eds.) (2014). Alkali-Activated Materials: State-of-the-Art Report, Rilem, T.C. 224-AAM, springer, Dordrecht, 11-57. DOI: 10.1007/978-94-007-7672-2_2

[6] Duxson, P., Fernández-Jiménez, A., Provis, J.L., Lukey, G.C., Palomo, A., Van Deventer, J.S.J. (2007) 'Geopolymer technology: The current state of the art', Journal of Materials Science, 42, 2917-2933. Available at: https://link.springer.com/article/10.1007/s10853-006-0637-z

[7] Luukkonen, T., Abdollahnejad, Z., Yliniemi, J., Kinnunen, P., Illikainen, M. (2018). One-part alkali-activated materials: A review. Cement and Concrete Research, 103, 21-34. DOI: 10.1016/j.cemconres.2017.10.001

[8] Nematollahi, B., Sanjayan, J., Qiu, J., Yang, E.-H. (2017). Micromechanics-based investigation of a sustainable ambient temperature cured one-part strain hardening geopolymer composite. Construction and Building Materials, 131, 552-563. DOI: 10.1016/j.conbuildmat.2016.11.117

[9] Hajimohammadi, A., van Deventer, J.S.J. (2017). Characterisation of one-part geopolymer binders made from fly ash. Waste and Biomass Valorization, 8, 225-233. DOI: 10.1007/s12649-016-9582-5

[10]. Suwan, T., Fan, M. (2017) 'Effect of manufacturing process on the mechanisms and mechanical properties of fly ash-based geopolymer in ambient curing temperature. Materials and Manufacturing Processes, 32, 461-467. DOI: 10.1080/10426914.2016.1198013

[11] Ouyang, S., Chen, W., Zhang, Z., Li, X., Zhu, W. (2020). Experimental study of one-part geopolymer using different alkali sources. Journal of Physics: Conference Series, 1605, 012155. DOI: 10.1088/1742-6596/1605/1/012155

[12] Nematollahi, B., Sanjayan, J., Shaikh, F.U.A. (2015). Synthesis of heat and ambient cured one-part geopolymer mixes with different grades of sodium silicate. Ceramics International, 41, 5696-5704. DOI: 10.1016/j.ceramint.2014.12.154

[13] Luukkonen, T., Abdollahnejad, Z., Yliniemi, J., Kinnunen, P., Illikainen, M. (2018) 'Onepart alkali-activated materials: A review', Cement and Concrete Research, 103, 21-34. DOI: 10.1016/j.cemconres.2017.10.001

[14] Fang, G., Ho, W.K., Tu, W., Zhang, M. (2018). Workability and mechanical properties of alkali-activated fly ash-slag concrete cured at ambient temperature, Construction and Building Materials, 172, 476-487. DOI: 10.1016/j.conbuildmat.2018.04.008.

[15] Rangan, B., Wallah, S., Sumajouw, D., Hardjito, D. (2006). Heat-cured, low-calcium, fly ash-based geopolymer concrete. Indian Concrete Journal 80(6), 47-52.

[16] Wallah, S., Rangan, B. (2006). Low-calcium fly ash-based geopolymer concrete: long term properties. Research Report G. C., Faculty of Engineering, Curtin University of Technology,

Perth, Australia. Available at: <u>https://www.researchgate.net/publication/265277484_Low-calcium_fly_ash-based_geopolymer_concrete_long_term_properties</u>

[17] Fernández-Jiménez, A., Palomo, A., López-Hombrados, C. (2006). Some engineering properties of alkali activated fly ash concrete. ACI Materials Journal, 103(2), 106–12, 2006.

[18] Tajunnisa, Y., Sugimoto, M., Sato, T., Shigeishi, M. (2017). A study on factors affecting geopolymerization of low calcium fly ash. International Journal of GEOMATE, 13, 100-107. DOI: 10.21660/2017.36.84153

[19] Tajunnisa, Y., Sugimoto, M., Uchinuno, T., Sato, T., Toda, Y., Hamasaki, A., Yoshinaga, T., Shida, K., Shigeishi, M. (2017) 'Effect of GGBFS and micro-silica on mechanical properties, shrinkage and microstructure of alkali-activated fly ash mortar', International Journal of GEOMATE, 13, 87-94. DOI: 10.21660/2017.39.11341

[20] Fernández-Jiménez, A., Palomo, J.G., Puertas, F. (1999). Alkali-activated slag mortars: Mechanical strength behaviour. Cement and Concrete Research, 29, 1313-1321. DOI: 10.1016/S0008-8846(99)00154-4

[21] Kurtoglu, A. E., Alzeebaree, R., Aljumaili, O., Nis, A., Gulsan, M. E., Humur, G., Cevik, A. (2018). Mechanical and durability properties of fly ash and slag based geopolymer concrete. Advances in Concrete Construction, 6(4), 345–362. DOI: 10.12989/acc.2018.6.4.345

[22] Mizuno, H., Iyoda, T. (2019). Shrinkage characteristics of ground granulated blast furnace slag high content cement. Integrated Vision – Caspeele, Taerwe & Frangopol (Eds) © 2019 Taylor & Francis Group, London, ISBN 978-1-138-62633-1

[23] Taghvayi, H., Behfarnia, K., Khalili, M. (2018). The Effect of Alkali Concentration and Sodium Silicate Modulus on the Properties of Alkali-Activated Slag Concrete. Journal of Advanced Concrete Technology, 16, 293-305. DOI:10.3151/jact.16.293

[24] Pundiene, I., Pranckeviciene, I., Zhu, Ch. (2020). Effect of Molarity and Temperature of Alkaline Activator Solution on the Rheological Properties and Structure Formation of Alkali-Activated Refractory Materials. Glass and Ceramics,77, 51–56. DOI: 10.1007/s10717-020-00236-1

[25] Mustata Al Bakri, A.M., Kamarudin, H., Bnhussain, M., Rafiza, A.R., Zarina, Y. (2012). Effect of Na2SiO3/NaOH Ratios and NaOH Molarities on Compressive Strength of Fly-Ash-Based Geopolymer. Aci Materials Journal, 109(5), 503-508.

[26] Hamzah, H.N, Abdullah, M.M. A.B., Yong, H., Rozainy, Z, Nor, A., Zailani, W.W. (2016). Correlation of the Na₂SiO₃ to NaOH Ratios and Solid to Liquid Ratios to the Kedah's Soil Strength. MATEC Web of Conferences. 78. 01071. DOI:10.1051/matecconf/20167801071.

[27] Tajunnisa, Y. (2017). Characterization of Alkali-activated Materials Cured at Ordinary Temperature for Utilization of PVA Fiber-Reinforcement. Doctoral thesis. Kumamoto University. Kumamoto. Japan.

[28] Kumar, S., Kumar, R., Mehrotra, S. P. (2010). Influence of granulated blast furnace slag on the reaction, structure and properties of fly ash based geopolymer. Journal of Materials Science, 45(3), 607-615. DOI:10.1007/s10853-009-3934-5

[29] Burciaga-Díaz, O., Escalante-García, J.I. (2017). Comparative performance of alkali activated slag/metakaolin cement pastes exposed to high temperatures, Cement and Concrete Composites, 84, 157-166. DOI: 10.1016/j.cemconcomp.2017.09.007

Chapter 4

Retardation Behaviour of Sucrose on Alkali-activated Materials

4.1 Introduction

In chapter 3, it was mentioned that the hardening of the mortar samples containing 40% or more GGBFS was very quick, so technically it can be said that they had a very short setting time. The result confirms the previous research [1,2]. The problem causes a workability problem as it was difficult to cast the fresh mortar into the molds. Therefore, it was decided to use an admixture to retard the setting process. Sucrose is an organic material. Its retardation influence on ordinary Portland cement materials was studied by several researchers in the past [3,4]. They concluded that sucrose has the ability to delay the setting process. Moreover, some of them also reported that besides the retarding effect, sucrose improves the microstructure of the OPC-based materials.

The retardation mechanism of the sucrose is not fully understood up to now, but it is generally accepted that sucrose adsorbs to the surface of hydrating particles to delay the production of the main hydrated material such as the C-S-H product [3]. Baruere [5] suggested that when mono-saccharides and disaccharides convert to the saccharinic acids by dilute alkali, these acids have retarding ability through adsorbing onto cement particles. Also, Milestone [6] has concluded that sugar and sugar acids adsorb onto the Ca ions which form tri-calcium silicate particles. So, blockage of hydration is occurring when the calcium silicate hydrate nucleation site surface is poisoned by the adsorption of sugar-acid anions.

This research aims to study the retardation effect of sucrose on AAM. The retardation effect of sucrose will be studied by the setting time test. Moreover, other properties of the AAM which have been studied in the previous experiment such as workability, compressive and flexural strength, and drying shrinkage test will be investigated in the presence and absence of sucrose.

4.2 Objective

This research will focus on the effect of sucrose on the initial and final setting time of AAM. Moreover, one of the mixes from the previous experiment which contains a higher amount of GGBFS will be chosen and 2% of the mass of aluminosilicate precursors, sucrose will be added as an admixture. The material's flowability, compressive and flexural strength and drying shrinkage will be studied, and the result will be compared with the one which doesn't have sucrose.

4.3 Experiment

4.3.1 Materials

Aluminosilicate precursors such as fly ash type F, GGBFS and silica fume, Alkaline activators (AA) such as NaOH pellets and granular Na₂SiO₃.5H₂O were used in this experiment. For the

mortars, one-part mixing and for the making of the paste two-part mixing was adopted. Standardized sand which conforms to JIS A 6207 was used as fine aggregate in mortar preparation. Sucrose was used as an admixture to retard the reaction process and to delay the solidification of the materials.

4.3.2 Mortar preparation and experimental program

Mortar	Aluminosilicate mix proportion (%)	PFB C (%)	Fly ash Type II (%)	GGB FS (%)	Silica fume (%)	Sucros e (%)	NaOH (mol/L)	AA ratio (g)	AA state
AAM-8	FA42.5/GGBFS42.5 /SF15	0	42.5	42.5	15	0	8	1.5	Solid
GM7	FA42.5/GGBFS42.5 /SF15	0	42.5	42.5	15	2	8	1.5	Solid

Table 4.1 Mix design of the mortar used in this experiment.

FA=fly ash, S=silica fume

The AAM-8 mortar which was prepared and tested in the previous experiment was chosen as a reference sample. The difference between AAM-8 and GM7 is the addition of sucrose. GM7 has sucrose by 2% of the mass of aluminosilicate precursors in its composition while AAM-8 is made without using sucrose.

For each type of mixture, four sets of 3 molds were prepared. 3 sets for the strength test (7, 14, 28, and 56 days), and one set for the shrinkage test.

The mixing procedure of both mixtures was the same as explained in the previous chapter. In GM7 the sucrose was mixed together with the aluminosilicate precursors and AA materials. Casting, vibrating, curing of the samples were the same as in the previous study.

The flowability, compressive and flexural strength and drying shrinkage of the samples were measured.

4.3.3 Paste preparation and setting time test

Paste samples were made to study the influence of sucrose on the setting time of alkali-activated materials. Table 4.3 shows the mix proportion of aluminosilicate precursors and composition characteristics of the materials.

Pastes	Aluminosilicat e mix proportion (%)	PFB C (%)	Fly ash Typ e II (%)	GGBF S (%)	Silica fume (%)	Sucros e (%)	NaOH (mol/L)	Sodium metasilic ate (%)	AA ratio (g)	AA state
P1	FA100	0	100	0	0	0	8	32	1.5	Solution
P1S	FA100	0	100	0	0	2	8	32	1.5	Solution
P2	FA80/GGBFS 20	0	80	20	0	0	8	32	1.5	Solution
P2S	FA80/GGBFS20	0	80	20	0	2	8	32	1.5	Solution

	Table 4.2 Mix	design and	composition	characteristics	of pastes.
--	---------------	------------	-------------	-----------------	------------

For the setting test time test, the maximum amount of the GGBFS was chosen to be 20%. Because the hardening of the paste and mortar with a higher percentage of GGBFS than the aforementioned percentage was occurring quickly soon after the mixing. This problem was obvious during testing the flowability. As for each type, two times flow measures were conducted, for the higher GGBFS based AAM, there was a significant difference of flow of each measure. Therefore, for studying the setting time, 100% fly ash-based and 20% slag-based AAM paste were chosen.

For making the pastes, an AA solution from NaOH pellets and Na₂SiO₃.5H₂O was made. The AA solution to the binder or aluminosilicate materials ratio was kept to 0.4 for the P1 and P1S, and 0.35 for the P2 and P2S paste.

A Vicat apparatus was used to test the setting time. Vicat apparatus consisted of units that were equipped with a frame and backlight platform, graduated scale, 10mm (0.4in) dia. plunger assembly, adjustable indicator, mold, glass plate and needle for testing the initial and final setting time.

For testing the initial setting time, a needle with a cross-sectional area of 1 mm^2 and for the final setting test, a needle with an annular (5mm diameter) attachment was used.

The mixing procedure was conducted according to the JIS R501. A paddle mixer was used to mix the materials. First, the aluminosilicate precursor and AA solution were mixed for 60 seconds. Then the mixing was stopped for 30 seconds, followed by the mixing for 90 seconds. The paste was then poured into the mold of the Vicat apparatus.

For the initial setting time test, the needle was lowered until it touched the surface of the AAM paste. Then the needle was gently released. This procedure was continued in different intervals of time until the needle was stopped 1 mm above the glass plate or the reading from the bottom of the scale was 1 mm. For the finding final setting time, the needle with an annular attachment was used. The needle was first lowered to contact the surface of the paste. Then it was released. The procedure was continued until the was no sign of the annular or circular attachment on the surface of the paste. The setting time was calculated from the starting period of the mixing until the time of the initial and final set was achieved.



Figure 4.1 Setting time test: (A) Vicat apparatus, (B) Paste with marks of the needles

4.4 Result

4.4.1 Setting time

Figure 4.2 shows the initial and final setting time result of the P1 and P1S pastes. The result shows that sucrose retarded the initial and final setting time of the pastes significantly. Sucrose increased the initial setting time of P1S from 310 minutes to 515 minutes and the final setting time from 433 minutes to 990 minutes.



Figure 4.2 Setting test result of the P1 and P1S

Chapter 4. Retardation Behaviour of Sucrose on Alkali-activated Material

Figure 4.3 shows the initial and final setting time to P2 and P2S. The result followed the same trend as the previous one. P2, which didn't have sucrose, had very short initial and final setting times of 10 and 30 minutes, respectively, but adding the 2% sucrose increased the paste's initial and final setting time to 23 and 48 minutes, respectively.



Figure 4.3 Setting test result of the P2 and P2S

The initial and final setting time test results of both kinds of pastes revealed that the retarding behavior of sucrose can also be observed on alkali-activated materials. As many researchers confirmed its retarding effect on ordinary Portland cement materials in the past [3,4].

4.4.2 Flowability

Figure 4.4 shows the flowability result of AAM-8 and GM7. The addition of 2% sucrose increased the flowability by 14%. AAM-8 had a flowability value of 128 mm while GM7 had 146 mm. This significant increase in flowability is attributed to the retarding behavior of sucrose. Because AAM-8's flow was quickly decreasing after the mixing time. But sucrose maintains the flow rate of the GM7 longer than the AAM-8.

Chapter 4. Retardation Behaviour of Sucrose on Alkali-activated Material



Figure 4.4 Flowability test result

4.4.3 Strength

Figure 4.5 and Figure 4.6 show the compressive and flexural strength results.

Sucrose increased the compressive strength. The compressive strength of G7, for 7, 28, and 56 days was increased by 30.5, 16.4, and 8% respectively. Importantly, sucrose had a significant influence on increasing the early compressive strength of the material.



Figure 4.5 Compressive strength result

The flexural strength result followed the same trend as the compressive strength. Sucrose improved the flexural strength as well. AAM-8 had a flexural strength of 4.6, 6.3, and 6.2 MPa for 7, 28, and

Chapter 4. Retardation Behaviour of Sucrose on Alkali-activated Material

56 days respectively, while GM7 had a strength of 4.6, 7.1, and 7.6 MPa for 7, 28, and 56 days respectively.

The increase in strength of the AAM using sucrose was due to their better flowability, which caused the aluminosilicate precursors and solid AA to better disperse through the mortar. Sucrose has the ability to improve the reaction process between the binders [3], this can also be a factor of the strength increment.



Figure 4.6 Flexural strength result

4.4.4 Drying shrinkage

Figure 4.7 shows the drying shrinkage of the AAM-8, G7, and OPC mortars. The shrinkage result of the OPC is included and used as a reference sample.

Sucrose had a significant influence on decreasing the drying shrinkage of the material. The drying shrinkage of GM7 was almost half of the AAM-8 and slightly lower than the OPC.

The higher drying shrinkage of AAM-8 was due to the larger amount of GGBFS in its composition which caused a high rate of hydration inside the mortar. But using sucrose significantly reduced the hydration intensity and improved the reaction process.





Figure 4.7 Drying shrinkage result

4.5 Conclusion

In the previous experiment, it was noticed that the mortars with higher percentages of GGBFS had a very quick setting time that creates difficulty in handling and casting them into the molds. Therefore, this research aimed to use sucrose as an admixture to retard the reaction process and delay the setting time. The following conclusion can be acquired from this study;

- 1. Using sucrose has a significant influence on the setting time of the AAM. It increases both the initial and final setting time of the materials.
- 2. Sucrose improves the flowability of AAM made by the one-part mixing method.
- 3. Both the compressive and flexural strength can be increased by using sucrose as an admixture in the AAM. The more significant effect can be seen in the increment of early compressive strength. The higher strength of the AAM with using sucrose is attributed to the better dispersion of the aluminosilicate precursors and AA all over the mortar.
- 4. AAM with sucrose has a much lower drying shrinkage than the one without sucrose. This is due to the reduction in the hydration rate and reaction process of AAM containing the admixture.

Reference

[1] Tajunnisa, Y. (2017). Characterization of Alkali-activated Materials Cured at Ordinary Temperature for Utilization of PVA Fiber-Reinforcement. Doctoral thesis. Kumamoto University. Kumamoto. Japan. Available at:

[2] Fang, G., Ho, W.K., Tu, W., Zhang, M. (2018). Workability and mechanical properties of alkali-activated fly ash-slag concrete cured at ambient temperature, Construction and Building Materials, 172, 476-487. DOI: 10.1016/j.conbuildmat.2018.04.008

[3] Juenger, M.C.G., Jennings, H.M. (2002). New insights into the effects of sugar on the hydration and microstructure of cement pastes. Cement and Concrete Research., 32, 393-399. DOI: 10.1016/S0008-8846(01)00689-5

[4] Ahmad, S., Lawan, A., Al-Osta, M. (2020). Effect of sugar dosage on setting time, microstructure and strength of type I and type V Portland cements. Case Studies in Construction Materials, 13. DOI: 10.1016/j.cscm. 2020.e00364

[5] Bruere, G.M. (1966). Set-retarding effects of sugars in Portland cement pastes. Nature, 212, 502–503. DOI: 10.1038/212502a0

[6] Milestone, N.B. (1979). Hydration of tricalcium silicate in the presence of lignosulfonates, glucose, and sodium gluconate. Journal of the American Ceramic Society, 62, 321-324. DOI: 10.1111/j.1151-2916. 1979.tb19068.x

Chapter 5

A Consideration on the One-part Mixing Method of Alkali-activated Materials: Property Comparison with the Two-part Mixing Method and Solubility Problem of Sodium Silicate

5.1 Introduction

Some researchers reported [1,2] that the AAM made by the one-part mixing method had lower mechanical strength than the one made by the two-part mixing method. Ouyang et al [1] found some undissolved particles of sodium metasilicate pentahydrate by studying the microstructure of the samples. Therefore, inappropriate dissolution of the alkaline source may cause the problem. Basically, the alkaline solution is the main part of geopolymer and AAM for geopolymerization and activation of the aluminosilicate precursors to make the binder [3]. Hence, there is a need for studying the solubility of both kinds of alkaline activators. NaOH pellets and granular Na₂SiO₃.5H₂O in different solvent temperatures. The result will help the further development of the method.

Granular sodium metasilicate pentahydrate was used for making AAM by one-part mixing method in the previous experiment, moreover, some other researchers also used the material while adopting the same mixing method [1,4,5]. This is maybe due to the higher solubility [6, 7] and availability of the material compare to the other types of solid sodium silicate. So, there is a need for studying the properties of sodium metasilicate aqueous solution and its influence on the characteristics of AAM. Because there isn't any study available regarding this. Moreover, this experiment characterizes the applicability of sodium metasilicate solution in a two-part mixing method. Traditionally, in a two-part mixing method, AAM is produced by using water glass gel that has a modulus (SiO₂:Na₂O) of around 2 which make the AAM viscose and cause application problem [8,9,10]. But sodium metasilicate has the modulus of 1. And according to Davidovits [11], the viscosity of the water glass is decreased when reducing the molar (SiO₂:Na₂O) ratio.

In the previous experiment, the authors observed the problem of the quick setting of the mortar which had a significant amount of slag in their composition, but using sucrose delayed the solidification of the mortar and facilitate the casting of the mortar. Therefore, sucrose was used as an admixture in the mortars which were produced to characterize the properties of the AAM made by Na₂SiO₃.5H₂O solution.

5.2 Objective

The objectives of this study can be defined as follow;

1. Comparison of the flowability and compressive strength of the samples made by a one-part and a two-part mixing method. Remember that the chemical composition of both types is the same.

2. Solubility test of sodium metasilicate pentahydrate ($Na_2SiO_3.5H_2O$) by a solvent with different temperatures.

3. Characterization of an aqueous solution made by sodium hydroxide (NaOH) and Na₂SiO₃.5H₂O.

4. Study the properties of mortars using Na₂SiO₃.5H₂O aqueous solution as a water glass.

5.3. Experiments

5.3.1 Material

The materials utilized were fly ash type II, slag with a Blaine size of $6000 \text{ cm}^2/\text{gr}$, and micro-silica as aluminum silicate precursors. Alkaline materials used were NaOH pellets and granular Na₂SiO₃.5H₂O. Sucrose was used as an admixture to improve the reaction process. Standardized sand conforming to JIS standard was used as fine aggregate. The chemical composition and other properties of the materials are explained in chapter 3 of the thesis. AA aqueous solution was made for making mortars by a two-part mixing method. The procedure of making an AA solution is explained in the next section.

5.3.2 Making alkali solution

The Water Bath Shaker was used to make an alkaline activator (AA) solution. It was equipped with a heater to control the temperature of the water inside, wires that can hold the flasks, and a shaking mechanism. For the making of the alkaline solution, firstly, the shaker's tub was half-filled with water. Then, flasks with a specified amount of distilled water were placed inside it and tightened by its wires. The temperature of the water inside the tub was set to 30°C. The frequency of the shaking was at 120 rounds/min. Specific NaOH pellets were put inside each of the flasks in two sets, then shook for 3 minutes for complete dissolution. Granular Na₂SiO₃.5H₂O were placed inside the flasks in 3 equal sets of shakings. During the dissolution of NaOH pellets in water, the solution's temperature increased. It helped the first set of granular Na₂SiO₃.5H₂O could not be dissolved well. Therefore, in the second and third sets, the tub water temperature was increased to 40 and 45°C, respectively.

After Na₂SiO₃.5H₂O was dissolved entirely in NaOH solution, they were put in a larger flask and stored at room temperature. After some while, it was noticed that the solution was susceptible at normal temperature, and crystallization of the solutes could quickly occur inside the solution. Therefore, the solution was put in an incubator at 45°C. After that, the crystallization of the solute disappeared, and the solution became transparent, smooth, not viscose, and looked like water



Figure 5.1. Making alkaline solution: (A, B) Water Bath Shaker, (C) AA solution after dissolution process, (D) Crystallization of the solutes, (E) An incubator, (F) AA solution after heat treatment in the incubator.

5.3.3 Comparison of the properties of samples made by the one-part and two-part mixing method

To compare the properties of the AAM mortars made by one-part and two-part mixing methods, low calcium-based fly ash was used as aluminosilicate precursor and 8M NaOH and 32% Na₂SiO₃.5H₂O were used as AA. The AA/ aluminosilicate materials ratio was set to around 0.6 for all the samples. For the one-part mixing method, the AA was used in the dry state while in the two-part mixing method, they were used in the solution state. The AA solution was made by using Water Bath Shaker.

Sample	Fly ash (%)	AA	Mixing Method
GO	100	NaOH+Na2SiO3.H20	One-part
GT	100	NaOH+Na2SiO3.H20	Two-part

Table 5.1 The specification of each type of mortar

For the one-part mixing method, the water temperature was set to 24 to 25°C. while for the twopart mixing method, the temperature of the AA solution was around 40°C, because as it was

mentioned in the previous section that crystallization could occur at a lower temperature. The mixing procedure was the same as explained in the previous chapters. For the mortars made by the one-part mixing method, they were vibrated just after the casting. But for the two-part mixing method, the flow was higher enough that vibration was not required. The samples made by the two-part mixing method were demolded after 48 hours but the sample made by the one-part mixing method took a long time for the consolidation and setting, therefore, they were demolded after 7 days. After the demolding procedure, they were sealed with plastic and were cured at around 20 °C in a controlled temperature room until their compressive strength test.

The flowability and the compressive strength properties of the mortars were studied and compared. The flowability was tested according to JIS R 5201. To study the 7-, 28- and 56-days' compressive strength, cylindrical samples were made. The test procedure was carried out according to JIS R 5201.

5.3.4 Experimental program for characterization of AAM made by sodium metasilicate pentahydrate solution

For this experiment, four types of mortar samples were made as shown in Table 1. NaOH solution's molarity was 8M. The concentration of the sodium metasilicate solution was 32%, and the ratio of the alkaline solutions (Na₂SiO₃: NaOH) was set to 1.5 for all types of mortar and paste samples.

Sample	Fly ash (%)	GGBFS (%)	Silica fume (%)		
M1	80	20	0		
M2	70	30	0		
M3	75	20	5		
M4	100	0	0		

Table 5.2 Percentages of the aluminum silicate precursors in each type of the sample

Mi x	Aluminosilicate materials			Alkali activator (AA) aqueous solution			Mixing Water (10% of aluminosilicat			
	Fly ash	GGBF S	Silica fume	Sand	NaO H	Na ₂ SiO ₃ .5H ₂ O	Solvent water	e materials)	Sucros e	L/B *
M1	422.8	105.7	0	1536.2	29.8	93.9	160.8	52.9	10.6	0.64
M2	373.9	160.2	0	1536.2	31.0	97.2	166.4	53.5	10.7	0.65
M3	397.9	106.1	26.5	1536.2	29.8	93.9	160.8	53.0	10.6	0.64
M4	517.6	0.0	0	1536.2	29.2	92.1	157.5	51.8	10.4	0.64

Table 5.3 Mixture proportions of AAM mortars (kg/m³)

*Liquid/binder. L: Solvent water + Mixing water (10% of aluminosilicate precursors) + NaOH + Na2SiO3.5H2O. Binder: Aluminosilicate materials.

5.3.4.1 Preparation of the aluminosilicate precursors

To improve the reaction process for making the AAM, fly ash, slag and silica were premixed in a mixer called OMNI mixer.

The authors used an Omni mixer with a 5L capacity and rotation speed of 2.5Hz (2.5 spin or round per second) to mix the materials. Three liters of the materials were premixed, and they were covered by plastic bags and put inside the bucket with a lid. The duration time of the mixing was first set to 2 hours. But when the materials were not appropriately mixed and some white slag particles were apparent, then the mixing duration time was increased to 3 hours. After the 3-hour mixing, the materials were well mixed than before.

5.3.4.2 Mixing, casting and curing

The experiment prepared prisms of 160x40x40 mm for compressive and flexural strength and shrinkage tests. The materials were mixed in a paddle mixer. The mixing procedure was the same as described in chapter 3.

The M2 mortar required 10% more water to have a flow with sufficient workability, so it was decided to use 10% added water to all samples. The 10% extra water was added together with standardized sand in the mixing process.

During the mixing procedure, as the solution's temperature decreased due to the solute crystallization, the flow and the workability of the mortars decreased. Therefore, it was too difficult to cast them in the molds. This mixing condition was called the ambient mixing condition.

To solve the workability problem, it was decided to keep the mixing materials at 45°C in the incubator for at least 24 hours. And then mix the materials at a constant temperature of 24-26 °C. This mixing condition was named a hot mixing condition. The crystallization of the solution was not observed in the hot mixing condition and the mortar had sufficient workability. The mixing procedure was done at the temperature above inside a tent. The study observed significant improvement in the flow and rheology of the mortar.

The flow rate was measured according to JIS (R5201). After that, the mixtures were poured inside prisms, then vibrated for 2 minutes. To avoid the crystallization of the solution during the curing and facilitate the reaction process between aluminum silicate precursors and the alkaline solution, the prisms were put at 45°C and 94% relative humidity inside an incubator for 2 days.

5.4 Result and discussion

5.4.1 Comparison of the properties of samples made by the one-part and two-part mixing method

5.4.1.1 Flowability

Figure 2 shows the flowability of the mortar made by the one-part and two-part mixing methods. In the two-part mixing method, the flow value of the mortar is 244 mm which is significantly higher than the one made by the two-part mixing method. The flow value of GO which is made by the one-part mixing method is 159 mm. The lower flow rate of the GO is due to the poor solubility of Na₂SiO₃.5H₂O, the detail of the solubility of the material is described in the forthcoming section. Whereas, in GT, the AA is in a solution state and all the materials were well dissolved in the solvent.



Figure 5.2 Comparison of the flowability

5.4.1.2 Compressive strength

The samples made by the two-part mixing method had significantly higher compressive strength than the samples made by the one-part mixing method at 7, 28, and 56 days. The compressive strength of the GT was higher than the GO by 104% at 7 days, by 357% at 28 days, and by 725% at 56 days' test. This indicates that the AA solubility had a serious and important effect on the properties of the materials. Furthermore, the AA solution's temperature in GT may also have an impact on the increasing strength of the AAM.


Figure 5.3 Comparison of the compressive strength

5.4.2 Solubility of the solid alkaline materials

The reaction between aluminosilicate precursors and alkali-activated materials is vital for producing Geopolymer and AAMs. Therefore, it is important that in a one-part mixing method solid alkaline materials such as NaOH and sodium silicate are well dissolved to achieve the material with the required properties.

A solubility test of the materials was conducted to better understand the alkaline materials' solubility mechanism in the one-part mixing. One batch mixing amount of NaOH pellets, granular sodium metasilicate pentahydrates, and water was put inside a flask. The Na₂SiO₃.5H₂O: NaOH ratio was set to 1.5. Water Bath Shaker was used to mix the materials. The shaking frequency was set to 120 rounds/min.

The temperature of the water before mixing with the alkaline materials was 26°C. NaOH pellets and Na₂SiO₃.5H₂O were put inside the flask and shook for 5 minutes. The duration of shaking was chosen according to the estimated mixing timing of making mortar or concrete in a mixer. After 5 minutes of mixing, the temperature of the solution increased to about 34°C. As the dissolution of NaOH in water is an exothermic reaction, it rose the temperature of the solution to about 8 degrees Celsius. The majority of the particles couldn't be dissolved in water and those particles were the sodium metasilicate since NaOH quickly dissolved in water, this was confirmed by another attempt, in which less than 5 minutes were required for the NaOH to be fully dissolved inside the solvent.

Therefore, the solution was mixed for two more minutes, but still, there was no improvement observed in the dissolution rate, and the temperature of the solution decreased to 31°C. Thus, to increase the temperature of the solution, the temperature of the water tub bath shaker was set to 35 °C and then to 40°C, and each was mixed for 5 minutes. The temperature of the solution increased to 32 and 36.5 °C respectively. A significant improvement in the dissolution rate of the

sodium metasilicate was observed. However, there were still some particles that remained undissolved. When the temperature of the solution was kept to 41 °C, total dissolution of the particles occurred.

This study shows that dissolution of the sodium metasilicate by a solvent with a lower temperature is difficult and couldn't be fully dissolved when adopting the drying mixing method to make geopolymer and alkali-activated materials. A temperature of around 40°C for the solvent is required to dissolve the material thoroughly. Concerning increment of some degrees of temperature by the NaOH, a solvent slightly lower than the temperature mentioned above can also be enough for solubility of the sodium metasilicate. The temperature of the aluminum silicate precursors and aggregates may also influence the solubility of the materials. Therefore, the hot mixing condition will have a significant effect on the one-part mixing method. Moreover, besides maintaining the proposed temperature, increasing the mixing duration and reducing the Na₂SiO₃.5H₂O: NaOH ratio may also play important role in the dissolution of the material in the one-part mixing method.

5.4.3 Characterization of AAM made by the sodium metasilicate pentahydrate solution

5.4.3.1 Flowability

This test was carried to observe the influence of the aluminosilicate precursors and change in mixing conditions on the flow of the samples. Moreover, the flow and rheology of the alkaline solution made by the sodium metasilicate are discussed in this section.

The alkaline solution made by NaOH pellets and sodium silicate pentahydrate was smooth and fluent and there was no sign of viscosity. But as mentioned before, the alkali solution was very sensitive to the temperature variation as the crystallization of the alkaline activator solution occurred at a temperature lower than 30°C. Therefore, the workability and flowability of the mortars were decreased at the aforementioned temperature. The same phenomena of crystallization of water glass can also happen while adopting the one-part mixing method using sodium silicate pentahydrate. The crystallization in the solution may reduce the reaction process between the alkali and the aluminosilicate precursors. Hence the properties of the alkali-activated product can be reduced.

In this study, it was decided to change the temperature of the materials and the mixing condition. All the materials were kept at 45 °C for 24 hours, and the mixing was done at a controlled temperature of 24-26 °C. When the mixing condition changed to the hot mixing condition, significant improvement in the workability of fresh mortar was observed and crystallization was disappeared.

Before applying the hot mixing condition, the M1 mix was made in an ambient mixing condition, and the flow of the mixing was so low that it couldn't be easily put inside the molds. The flow value of the M1 was 102.5mm. There was about a 15% increment in the flow value of the mixture when the hot mixing condition was applied. A higher flow rate was achieved due to a reduction in the mix's cohesiveness at a higher temperature.

The M4 (fly ash 100%) sample had the highest flowability among the samples, and its flow value could not be measured due to its overflow from the flow table. Figure 4 shows the flow of M4. While some studies [8,9,10] show low workability and high viscosity for fly ash-based geopolymer using Na-based silicate as an activator. They also reported that admixtures such as superplasticizers did not significantly improve the flowability and rheology of paste or mortar. However, this experiment shows that all the fresh mortars were smooth, fluent, and without viscosity. This was achieved due to the use of sodium metasilicate and adopting hot mixing conditions.



Figure 5.4 Flow value of M4

The slag and micro-silica decreased the flowability. The mortar with 30% of the slag had the lowest flow value of 114 mm. M3 and M1 had a flowability of 122 and 120 mm, respectively.



Figure 5.5 Flow value (mm)

5.4.3.2 Compressive and flexural strength

Compressive and flexural strength is an important characterization of AAM. As sodium silicate improves the reaction process between the precursors and alkali [12], the result of compressive and flexural will reveal if a reasonable early strength is achieved by using the sodium metasilicate. Therefore, each sample type was tested for compressive and flexural strength at 7, 28, and 56 days. The test used a Universal Testing Machine (UTM) with a capacity of 200 tons.



Figure 5.6 The compressive strength (MPa)

Slag had a significant influence on the compressive strength of the samples. M2, which had 30% slag, had the highest compressive strength at the age of 7, 28, and 56 days which are 39.2MPa, 39.6 MPa, and 43.4MPa, respectively, followed by M3 and M1.

M4, made of 100% fly ash, had the lowest compressive strength of all ages. M4 had a compressive strength of 17.8 MPa, 20.6MPa, and 21.2MPa for the ages of 7, 28, and 56 days respectively. The flow of M4 was much higher than the other types, decreasing the liquid/binder ratio can improve the strength of M4.

In accordance with the compressive strength result, M2 had the highest 7, 28, and 56-day flexural strength, which is around 6MPa—followed by M1 and M3, which had the flexural strength of around 5 MPa for all ages. In comparison, M4 had a flexural strength of 3.7 MPa for seven days, 4.2 MPa for 28 days, and 4.7 MPa for 56 days.



Figure 5.7 The flexural strength (MPa)

All the samples achieved reasonable early compressive and flexural strength by the age of 7 due to the Ca element's presence in the slag [6] and the influence of the sodium metasilicate as a source of silica.

Both compressive strength of more than 40 MPa and flexural strength of 6 MPa can be achieved by using 32% concentrated sodium metasilicate solution, which is much lower than the concentration of water glass or sodium silicate solution used in making traditional geopolymer and alkali-activated materials [13,14,15,16,17].

5.4.3.3 Drying shrinkage

Drying shrinkage of the samples was studied according to [18]. The test was conducted to study the influence of the mix proportion of the aluminosilicate precursors activated by the alkaline activator solution made by the sodium metasilicate and sodium hydroxide on the drying shrinkage property of the AAM.

M3, M2, and M1 had higher amounts of shrinkage compared to M4. It was due to the presence of the slag and silica fume in their composition. Which caused the quick reaction and early loss of water in the mixture. The shrinkage of samples containing slag was increased until 50 days, after that, it was decreased and then flattened. On the other hand, the shrinkage of the 100% fly ash sample was nearly 0 microstrain at the age of around 90 days. This is attributed to the slow reactivity of the precursor.

The shrinkage of all samples was below 0.07 percent, within the range of allowable shrinkage defined by the ACI 209R, which is 0.05 to 0.078% [19]. This is maybe due to the curing of the samples at 45°C in the incubator. Which conforms to the result of [17].



Figure 5.8 Shrinkage test result

5.5 Conclusion

One-part mixing method was studied by many researchers to make ease in the production of alkaliactivated materials and geopolymers and to generalize the materials. But some researchers found the quality of the materials made by the one-part mixing method is lower than the one made by the

two part-mixing methods. Therefore, this research reconfirms the properties of materials made by both kinds of mixing methods. Furthermore, none of the researchers checked the solubility of the alkaline materials used in the aforementioned mixing method. Therefore, this research studied the dissolution mechanism and solubility of the alkaline materials used in a one-part or dry mixing method. Moreover, the aqueous solution from sodium metasilicate and NaOH were made to overcome workability and rheological problems. This study concludes as follow:

- 1. Granular sodium metasilicate pentahydrate cannot be dissolved totally in a solvent with a temperature of around 25°C in 5 or more minutes of mixing. The problem of solubility can decrease the mechanical strength and other properties of AAM.
- 2. The dissolution of the sodium metasilicate increases with an increase in temperature. Total dissolution of the material occurs at a temperature of around 40°C. Hot mixing conditions can play a significant role in the material's solubility in the one-part mixing method.
- 3. The alkaline solution made by the sodium metasilicate is more fluent and smoother. This is due to the lower amount of SiO2 and concentration of the water glass. But the alkaline solution is more sensitive to the lower temperature. Crystallization of the solution can easily happen at a temperature below 30°C. The same phenomena of crystallization of water glass can also occur while adopting a dry mixing method. This can reduce the reaction process between the alkali and the aluminum silicate precursors and reduce the properties of the alkali-activated product. To avoid crystallization and to retain it smooth, keeping the solution at a higher temperature is recommended.
- 4. Mixing materials at the ambient condition method decreases the flow rate of the fresh mortar made by the sodium metasilicate solution due to the sensitivity of the solution at a lower temperature. Therefore, the hot mixing condition improves the flow rate of mortars
- 5. AAM samples made by the sodium metasilicate solution can acquire shape according to the molds, and the surfaces of the samples are the same as other types of alkali-activated material made by the traditional method.
- 6. Incorporating slags in AAM material has advantages such as improving the reactivity between aluminum silicate precursors and modifying the compressive and flexural strength and density. A compressive of around 40 MPa can be achieved by combined fly ash and slag-based AAM activated by a low concentrated sodium metasilicate and NaOH solution.
- 7. Using sodium metasilicate solution in the two-part mixing method can solve the viscosity problem created by sodium silicate gel.

Reference:

[1] Ouyang, S., Chen, W., Zhang, Z., Li, X., Zhu, W. (2020). Experimental study of one-part geopolymer using different alkali sources. Journal of Physics: Conference Series, 1605, 012155. DOI: 10.1088/1742-6596/1605/1/012155

[2] Nematollahi, B., Sanjayan, J., Shaikh, F.U.A. (2015). Synthesis of heat and ambient cured onepart geopolymer mixes with different grades of sodium silicate. Ceramics International, 41, 5696-5704. DOI: 10.1016/j.ceramint.2014.12.154

[3] Davidovits, J. (2011). Geopolymer Chemistry and Applications' 3rd ed., SaintQuentin. Available at:

https://www.researchgate.net/publication/265076752_Geopolymer_Chemistry_and_Applications. Institut Géopolymère

[4] Husin, N.A., Bayuaji, R., Tajunnisa, Y., Darmawan, M.S., Suprobo, P. (2020). Performance of high calcium fly ash based geopolymer concrete in chloride environment. International Journal of GEOMATE, 19, 107-113. DOI: 10.21660/2020.74.56966a

[5] Yang, K.H., Song, J.K., Ashour, A.F., Lee, E.T. (2008). Properties of cementless mortars activated by sodium silicate. Construction and Building Materials, 22, 1981-1989. DOI: 10.1016/j.conbuildmat.2007.07.003

[6] HERA (2005). Soluble silicates. Human & Environmental Risk Assessment on ingredients of European household cleaning products. Available at: <u>https://www.heraproject.com/files/14-F-05-RA%20Risk%20Assessment%20of%20Soluble%20Silicates%20final%20draft.pdf</u>

[7] Sodium metasilicate, RESOURCE CHEMICAL LTD. Available at: https://www.resourcechemical.co.uk/sodium-metasilicate.html

[8] Provis, J.L. (2009). Activating solution chemistry for geopolymers. In J.L. Provis, J.S.J. van Deventer (Eds.). Woodhead, Cambridge, UK. pp. 50-71. DOI: 10.1533/9781845696382.1.50, Geopolymers: Structures, Processing, Properties and Industrial Applications

[9] Xie, J., Kayali, O. (2016). Effect of superplasticiser on workability enhancement of Class F and Class C fly ash-based geopolymers. Construction and Building Materials, 122, 36-42. DOI: 10.1016/j.conbuildmat.2016.06.067

[10] Criado, M., Palomo, A., Fernández-Jiménez, A., Banfill, P.F.G. (2009). Alkali activated fly ash: Effect of admixtures on paste rheology. Rheologica Acta, 48, 447-455. DOI: 10.1007/s00397-008-0345-5

[11] Davidovits, J. 'Poly(siloxonate) and polysilicate, soluble silicate, Si: Al=1', In J. Davidovits
(Ed.). pp. 79-113 (2011), GEOPOLYMER Chemistry and Applications. Saint-Quentin, France Available

https://www.researchgate.net/publication/265076752_Geopolymer_Chemistry_and_Applications.

[12] Palomo, A., M. W. Grutzeck, M.T. Blanco. (1999). Alkali Activated Fly Ashes, A Cement for the Future. Cement and Concrete Research. 29(8), 1323-1329. https://doi.org/10.1016/S0008-8846(98)00243-9

[13] Tajunnisa, Y., Sugimoto, M., Sato, T., Shigeishi, M. (2017). A study on factors affecting geopolymerization of low calcium fly ash. International Journal of GEOMATE, 13, 100-107. DOI: 10.21660/2017.36.84153

[14] Tajunnisa, Y., Sugimoto, M., Uchinuno, T., Sato, T., Toda, Y., Hamasaki, A., Yoshinaga, T., Shida, K., Shigeishi, M. (2017). Effect of GGBFS and micro-silica on mechanical properties,

shrinkage and microstructure of alkali-activated fly ash mortar. International Journal of GEOMATE, 13, 87-94. DOI: 10.21660/2017.39.11341

[15] Chi, M., Chang, J., Huang, R. (2012). Strength and Drying Shrinkage of Alkali-Activated Slag Paste and Mortar. Advances in Civil Engineering, vol. 2012. DOI: 10.1155/2012/579732

[16] Nath, P., Sarker, P.K. (2014). Effect of GGBFS on setting, workability and early strength properties of fly ash geopolymer concrete cured in ambient condition. Construction and Building Materials, 66, 163-171. DOI: 10.1016/j.conbuildmat.2014.05.080

[17] Marjanović, N., Komljenović, M., Baščarević, Z., Nikolić, V., Petrović, R. (2015). Physicalmechanical and microstructural properties of alkali-activated fly ash-blast furnace slag blends. Ceramics International, 41, 1421-1435. DOI: 10.1016/j.ceramint.2014.09.075

[18] JIS (2010) A 1129-3 Method for Measurement for Length Change and Mortar and Concrete

[19] Tarr, S. (2012). Concrete mixture shrinkage potential', Concrete Construction. Available at: https://www.concreteconstruction.net/how-to/materials/concrete-mixture-shrinkage-potential_o

Chapter 6

A Study on The Influence of Sodium Silicate Concentration and Modulus Ratio, Temperature of the Raw Materials, Sucrose and Curing Conditions on the Properties of Low-Calcium Fly Ash-Based Alkali-activated Materials

6.1. Introduction

Low calcium fly ash-based AAM has some extraordinary advantages. They possess geopolymeric characteristics such as low shrinkage and high fire resistance [1]. But compared to the slag-based AAM they have lower strength [2,3] and longer setting time [4] when cured at ambient temperature conditions. The high setting time of the material is considered one of the major problems, this means that they can't set or harden well at ambient temperature curing conditions. They sometimes need more than 40 hours to achieve their final setting [4].

Incorporating the slag with fly ash improves the reactivity of the aluminosilicate precursors that result in hardening of the samples at ambient temperature and modify some other properties of concrete such as compressive strength, flexural strength, and density of the AAM [5]. As the slag is very active and increases the reactivity and hydration of the materials [6], this causes the materials to emit a higher amount of heat during the reaction process [7]. Therefore, the drying shrinkage of alkali-activated materials with slag is much higher than the other types [5, 3].

This research aims to solve the problem related to low calcium fly ash-based AAM. Therefore, the influence of the concentration and molar ratio of sodium silicate, etc. is studied on the setting time, flowability, and compressive strength of AAM.

Conventionally, for the production of fly ash-based AAM or for other types, sodium silicate is used at a concentration of around 50% [8, 9, 10] In the previous experiment while characterizing the one-part mixing method of AAM, granular sodium metasilicate was used with a concentration of 32%. A quick setting of the mortar consisting of 100% low-calcium fly ash was observed. Hence, the role of concentration and types of sodium silicate on the setting time of the AAM is considered to be studied in this experiment.

6.2. Objectives

Study the influence of the following factors on the setting time, flowability and compressive strength of the low-calcium fly ash-based AAM:

- 1. Sodium silicate molar ratio (SiO₂:Na₂O ratio).
- 2. Sodium silicate concentration.
- 3. The addition of sucrose
- 4. Raw material (fly ash and sand) temperature

5. Curing condition.

6.3. Experiment

6.3.1 Materials

Fly ash type II was used as an aluminosilicate precursor, the composition and other properties are illustrated in Table 1. Sodium silicate and sodium hydroxide were used as alkali activators (AA). The concentration of NaOH was 8 molars. Sodium silicate: NaOH ratio was set to 1.5 for the paste and mortar samples. According to the SiO₂:Na₂O modulus, two types of sodium silicate were used: Sodium metasilicate pentahydrate (Na₂SiO₃.5H₂O) which has the modulus of 1, and sodium disilicate (Na₂O.2SiO₂) with the SiO₂:Na₂O ratio of around 2. The sodium metasilicate was in the granular solid state, while sodium disilicate was used in form of solution gel. The granular sodium metasilicate consisted of 28% SiO₂, 28% Na₂O, and 54% H₂O, and sodium disilicate aqueous solution was composed of 36% SiO₂, 18% Na₂O, and 46% H₂O. Standardized sand conforming JIS R 5201 was used as a fine aggregate for making the mortars. Sucrose was used as an admixture in this experiment.

Table 6.1 Specific gravity (S.G.), the chemical composition of materials in weight percentage (wt. %) by XRF, and chemical analysis of standardized sand

Materials		Chemical composition (% weight)							
	SiO ₂	Al_2O_3	Fe ₂ O ₃	CaO	Na ₂ O	H ₂ O	K ₂ O	SG(g/cc)	
Fly ash	55.19	25.35	7.57	4.06				2.33	
Standardized	98.4	0.41	0.36	0.16	0.01	< 0.2	0.01	2.64	
sand									

6.3.2 Mix proportion and samples detail

In this research, 7 types of paste and 11 types of mortar samples were made. Table 2 and Table 3 show the chemical composition and other characteristics of the pastes and mortars respectively. AA solution was composed of NaOH and sodium silicate.

To investigate the effect of the sodium silicate solution (water glass) on the setting time, flowability, and compressive strength of the AAM, 3 patterns of sodium silicate solution were utilized. Sodium metasilicate solution with a concentration of 32% and sodium disilicate solution with concentrations of 32% and 54% were prepared. Each type of aforementioned solution was then mixed with 8M NaOH to make AA solutions. Granular sodium metasilicate pentahydrate was dissolved in water to make the solution. The procedure was conducted by a Water Bath Shaker. Sodium disilicate solution (a commercial name) with 54% concentration which is available commercially was provided, and for the making of 32% concentration, the estimated amount of water was added to the solution. In the previous experiment, the solubility of Na₂SiO₃.5H₂O was investigated. It was found that the granular sodium metasilicate couldn't be dissolved well in a solvent with a temperature below 40°C. Moreover, below the aforementioned temperature, the solution was sensitive and crystallization of the solute could easily occur. So, it was recommended

to use water or a solvent at around 40°C temperature while adopting a one-part mixing method. Therefore, for better characterization, all the AA solutions were used at 40°C in the experiment.

To study the influence of the temperature of the mixing materials on the properties of the AAM, the materials were used in two different temperature conditions, hot conditions and ambient conditions. In hot conditions, all the materials were used at 40 °C. But at the ambient conditions, except for the AA solution, the temperature of the raw materials such as fly ash and sand was around 20°C. There was a difference of 20 °C between those two mixing conditions. As mentioned before, AA solution was used at a temperature of around 40 °C both for ambient and hot mixing conditions.

The effect of sucrose on the properties of the paste and mortars was also studied. The amount of sucrose was set to 2% by the mass of fly ash.

For the characterization of the curing condition of the mortars, they were cured at ambient conditions and in an oven or incubators. At ambient condition, they were put at a controlled temperature room where the temperature was set to around 20 °C, but for the others which were cured inside incubators, the temperature was set to 45° C and was kept for 24 hours and then they were put inside the controlled temperature room until the compressive strength test day.

N o	Sample s	Fly ash (g)	SiO ₂ / Na ₂ O	32 % Sodium metasilicate concentrated +8M NaOH AA Solution (g)	32 % Sodium disilicate concentrated +8M NaOH AA Solution (g)	54 % Sodium disilicate silicate concentrated +8M NaOH AA Solution (g)	Sucros e (g)	Solution/Fl y ash	Raw material (fly ash, sand) temperat ure
1	PA	400	2	0	0	166	0	0.42	20°C
2	PH	400	2	0	0	166	0	0.42	40°C
3	P1A	400	1	140	0	0	0	0.35	20°C
4	P1H	400	1	140	0	0	0	0.35	40°C
5	P2A	400	2	0	140	0	0	0.35	20°C
6	P2H	400	2	0	140	0	0	0.35	40°C
7	P3A	400	1	140	0	0	8	0.35	20°C

Table 6.2 Mix	proportion	and detail	of each	paste type
---------------	------------	------------	---------	------------

No	Fresh mortar	Cylindri cal samples	Fly ash (kg/m ³)	Sand (kg/m ³)	Si ₂ O:N a ₂ O	32 % Sodium metasilicate concentrated+ 8M NaOH AA Solution (kg/m ³)	32 % Sodium disilicate concentrated+ 8M NaOH AA Solution (kg/m ³)	54 % Sodium disilicate silicate concentrated+ 8M NaOH AA Solution (kg/m3)	Sucrose (kg/m ₃)	Solution /Fly ash	Fly ash and sand temperat ure	Curing Conditio n
1	GA	GAO	517.6	1536.2	2	0	0	278.7	0	0.54	20°C	0
2	GH	GHO	517.6	1536.2	2	0	0	278.7	0	0.54	40°C	0
3		G1AA	517.6	1536.2	1	278.7	0	0	0	0.54	20°C	А
4	G1A	G1A0	517.6	1536.2	1	278.7	0	0	0	0.54	20°C	0
5		G1HA	517.6	1536.2	1	278.7	0	0	0	0.54	40°C	А
6	G1H	G1HO	517.6	1536.2	1	278.7	0	0	0	0.54	40°C	0
7		G2AA	517.6	1536.2	2	0	278.7	0	0	0.54	20°C	А
8	G2A	G2AO	517.6	1536.2	2	0	278.7	0	0	0.54	20°C	0
9		G2HA	517.6	1536.2	2	0	278.7	0	0	0.54	40°C	A
10	G2H	G2HO	517.6	1536.2	2	0	278.7	0	0	0.54	40°C	0
11	G3A	G3AA	517.6	1536.2	1	278.7	0	0	10.4	0.54	20°C	A

Table 6.3 Mix	proportion ar	d the details	of each mortar	type
---------------	---------------	---------------	----------------	------

Note: A= ambient temperature condition, O= Curing at 45°C

6.3.3 Paste preparation and setting time test

Preparation of the paste and setting time test was done according to the JIS R 5201[25]. For the mixing, a paddle mixer was used. First, fly ash and AA solution were mixed for 1 minute. Then the mixing was stopped for 90 seconds followed by a 1-minute mixing. Next, they were poured into a mold. The pouring procedure was required to be completed within a minute. Then initial and final setting times of the pastes were tested.

6.3.4 Mortar preparation, flow and compressive strength test

The fly ash was first mixed with the alkaline solution for 1 minute, followed by the addition and mixing of standardized sand for 30 seconds each at low and high speed. Then the mixing procedure was stopped for 90 seconds and the next high speed mixed for 1 minute.

Except for samples containing 54% sodium disilicate, all other mortar had enough flow and smoothness that didn't require a vibrator for the compaction. Samples with a 54% concentration of sodium disilicate were compacted by a vibrator.

Two patterns of curing were applied. At ambient condition and inside incubator. As mentioned before, for the ambient condition, the samples were cured inside a controlled temperature room in which the temperature was kept to around 20 °C. The other types were cured inside incubators for 48 hours and then cured at the controlled temperature room until their compressive strength test.

Flow test was conducted twice and the procedure was done according to the JIS R 5201 [11]. Cylindrical molds with a diameter of 5 cm and a height of 10 cm were used to study 7-, 28-, and 56-days' compressive strength. The compressive strength test was also conducted according to JIS R 5201[30].

6.4 Result

6.4.1 Influence of the sodium silicate concentration

In this experiment, sodium disilicate solution was used in two concentrations, 32 and 54%. In PA and GA, the concentration of the water glass was 54% while in P2A and G2A it was 32%.

6.4.1.1 Setting time

Figure 1 shows the setting time of AAM with different concentrations of sodium silicate. PA which was made by a conventional method had 54% water glass in its composition. The initial and final setting time of PA was around 35 hours, which was notably high and the result confirmed the previous research [4]. This type of AAM is not applicable on site. The slow reactivity of low-calcium fly ash in an alkaline environment is the factor behind the delayed setting time. Therefore, they need to be cured at higher temperatures or partially replaced with slag to reduce the setting time. But using slag increase the shrinkage of the material. Interestingly, when the concentration of the water glass was reduced to 32%, there was a significant development in the initial and final setting time of the paste. The initial and final setting time of the P2A was 5.8 and 7 hours respectively. The increase in the amount of water might improve the reactivity of the fly ash and cause the paste to set quickly.



Figure 6.1 Influence of the change in concentration of sodium silicate solution on the setting time of the pastes

6.4.1.2 Flowability

Figure 2 shows the effect of the sodium silicate concentration on the flow properties of AAM. GA, a mortar that was made of 54% sodium silicate has a flow value of 145 mm which was much lower than the 32% sodium silicate concentrated AAM. G2A had a flow rate of 250 mm. As mentioned in the introduction section, conventional alkali-activated mortar and concrete have a higher viscosity. This viscosity is caused by the presence of SiO_2 in the sodium silicate solution. In this experiment, increasing the water amount in the solution reduced the viscosity and improved the flowability of the fresh mortars.



Figure 6.2 Influence of the change in concentration of sodium silicate on the flowability of AAM

6.4.1.3 Compressive strength

Figure 6.3 shows the influence of the water glass concentration on the AAM mortar compressive strength. GAA and G2AA were made by 54 and 32% sodium disilicate solution respectively. Both of them were cured at ambient temperature conditions. In this study, GAA which had almost the same composition as GA of this experiment was taken from another research [10]. The result shows that G2AA with 25.8 MPa, significantly had higher compressive strength than the GAA. The compressive strength of the GAA was 8 MPa.

Using sodium silicate with the lower concentration decreased the setting time and increased the compressive strength of the AAM in ambient temperature conditions. This indicates that the reactivity of the fly ash was increased with an increase in the amount of water or a decrease in the PH value of the alkaline solution.



Figure 6.3 Influence of the change in concentration of sodium silicate on the compressive strength of AAM

6.4.2 Influence of the SiO₂:Na₂O ratio

6.4.2.1 Setting time

Figure 6.4 shows the influence of the sodium silicate molar ratio (SiO₂:Na₂O) on the initial and final setting time of P1A and P2A samples which was made by each 32% sodium metasilicate and sodium disilicate solution respectively. P1A had an initial setting time of 6.3 hours and a final setting time of 8.4 hours. Both the initial and final setting time of P1A was slightly higher than P2A which had initial and final setting times of 5.8 and 7 hours respectively. This indicates that the SiO₂:Na₂O ratio of sodium silicate solution with the same concentration doesn't have any significant influence on the setting properties of low-calcium fly ash-based AAM. The final setting time of both kinds of material was slightly lower than OPC paste. OPC paste has a final setting

time of around 10 hours [26]. This means that in terms of the setting time, both kinds of materials are reasonable for the cast-in-situ condition.



Figure 6.4 Influence of the SiO₂:Na₂O ratio on the setting time

6.4.2.2 Flowability

Same as the setting time test, the change in the SiO₂:Na₂O ratio didn't have any significant effect on the workability and flow rate of both types of AAM. Both types, G1A and G2A had good flow rates of 244 mm and 250 mm respectively.



Figure 6.5 Influence of the SiO₂:Na₂O ratio on the flowability

6.4.2.3 Compressive strength

Figure 6.6 shows the influence of the SiO₂:Na₂O on the compressive strength of both types of AAM. G1AA had a compressive strength of 4.7 MPa, 6.4 MPa, and 23.2 MPa at 7,28 and 56 days respectively. Whereas G2AA's 7-, 28- and 56-days' compressive strength were 4 MPa, 25 MPa, and 46 MPa respectively. This indicated that both types had almost the same compressive strength at 7 days but for 28 and 56 days, the compressive strength of the G2AA was increased by 403 and

719% respectively. This revealed that the increase in SiO₂ content of sodium silicate solution improved the compressive strength remarkably at ambient temperature curing.

This result indicates that with using 32% sodium disilicate, remarkable 28 and 56-day compressive strength can be achieved, which is appropriate for use in the structural field. However, the 7-day compressive was much lower, but that may be increased by a slight increase in the concentration or reducing the solution: fly ash ratio, as the flow value of the mortar, was high enough for the reduction.



Figure 6.6 Influence of the SiO₂:Na₂O ratio on the compressive strength

6.4.3 Influence of using sucrose

6.4.3.1 Setting time

There was a significant increment in the initial and final setting times of the AAM when sucrose was added to the mixture. P1A had the initial and final setting time of 5.2 and 8.6 hours respectively, but when sucrose was added to the mixture the initial and final setting times were increased to 7.2 and 16.5 hours respectively. Figure 6.7 shows the effect of sucrose on the initial and final setting of low-calcium fly ash-based AAM.



Figure 6.7 Influence of sucrose on the setting time

6.4.3.2 Flowability and compressive strength

In chapter 4 while characterizing the effect of sucrose on AAM made by the 42.5% fly ash, 42.5% slag, and 15% slag, it was found that adding 2% sucrose to the mixture increased the flowability and compressive strength. As the mixture without the sucrose had a quick setting, adding the admixture delayed the setting time, therefore, the mixture maintained its initial flow, and this cause the fly ash and other materials to disperse well through the mixture. Hence the flow value and compressive strength of the AAM with sucrose were higher than the AAM without sucrose. But in this experiment 100% fly ash was used and the flowability of the mortar was better after the mixing procedure, and quick setting did not occur. Therefore, sucrose didn't have a positive influence on the flowability and compressive strength of the AAM, instead, the flow value and the compressive strength were decreased with the addition of sucrose.



Figure 6.8 Influence of sucrose on the flowability



Figure 6.9 Influence of sucrose on the compressive strength

6.4.4 Influence of the temperature of the raw materials

To study the influence of the temperature of raw materials such as fly ash and sand on the properties of the AAM. They were studied in two different temperatures, an ambient temperature which was around 20°C, and hot conditions. In hot conditions, fly ash and sand were used at temperatures around 40° C.

6.4.4.1 Setting time

Figure 6.10 shows the influence of the fly ash and sand temperature on the initial and final setting time of all three types of AAM.

The temperature of the materials didn't have any significant influence on the initial and final setting time of the conventional type of fly ash-based AAM. P2H followed the same trend, the hot condition or temperature of the mixing materials did not decrease the setting time of the AAM. But, conversely, the final setting time of the P2H was increased by almost 1 hour. This result indicated that the temperature of the materials didn't have any obvious effect on the setting properties of AAM using sodium disilicate that has a SiO₂:Na₂O ratio of 2.

But the temperature of the materials and hot mixing method had a significant influence on the setting time of AAM using sodium metasilicate. The initial and final setting times of P1A were 6.3 and 8.4 hours respectively and after heating the material at an incubator, the initial setting and final setting times were decreased to 2.6 and 6.9 hours respectively. There was a decrease of 58% for the initial setting time and 18% for the final setting time.



Figure 6.10 Influence of the temperature of the raw materials on the setting time of AAM

6.4.4.2 Flowability

The hot mixing condition or preheating of the materials in the incubator increased the flow value of conventional fly ash-based AAM. GAO had a flow of 145mm and the other one (GHO) in which the aluminosilicate precursor and the fine aggregate were heated before the mixing had a flowability of 157 mm. This improvement of the flow was due to the breaking of the sodium silicate solution viscosity by the application of heat. The viscosity of the GHO was lower than the GAO. This result can help to overcome the viscosity problem of AAM which is created by the alkaline solution. Some researchers tried to solve the workability problem using some admixture, but none of them had apparent and considerable influence to reduce the viscosity and solve the workability problem [12, 13, 14].

G1AA and G2AA both had better workability and the viscosity had been already reduced by a decrease in the concentration of sodium silicate solution. Application of hot mixing conditions or heating of the mixing material decreased the flowability of both kinds of AAM. G1AA and G2AA had flow rates of 244 mm and 250 mm respectively, but after the application of the heat, the flow rates were decreased to 236 mm and 232 mm respectively. The decrease in the flow rate was due to the activation and increase in the reactivity of the fly ash by the application of the heat. This phenomenon was also confirmed by the compressive strength result.



Figure 6.11 Influence of the temperature of the raw materials on the flowability

6.4.4.3 Compressive Strength

Figure 9 shows the influence of the aluminosilicate precursor and fine aggregate temperature on the compressive strength of the fly ash-based AAM.

There was a slight increase in the compressive strength of the conventional AAM by applying heat on the mixing material. GAO had a compressive strength of 96.2 MPa at 28 days, while the compressive strength was increased to 99.9 MPa after the application of heat to the materials. The curing of both GAO and GHO was done at 45°C in an incubator.

The same trend of the conventional AAM result was followed by the G2HA, there wasn't any significant improvement in the compressive strength of the material at 7, 28, and 56 days.

However, there was a decrease in the strength of the AAM made by the 32% sodium metasilicate at 7 days, but there was a remarkable improvement of 34% at 28 days and an increase of 39% at 56 days' compressive strength of the aforementioned material.

The result clearly indicated that the compressive strength sodium disilicate was less influenced by the heat than AAM made by the sodium metasilicate. The finding confirmed the setting time result.





6.4.5 Effect of the curing condition on compressive strength

The heat curing had a significant influence on the 28-day compressive strength of AAM made by 54% sodium disilicate. Because of this, the conventional type of low-calcium fly ash AAM is cured at a higher temperature. At ambient curing conditions, GAA had a compressive strength of 8MPa at 28 days, and with the curing at 45°C for two days, the compressive strength was increased by around 12 times.

Moreover, the heat curing had a remarkable effect on the 7, 28, and 56-day compressive strength of the AAM made by the 32% sodium metasilicate. When applying the heat curing at 45°C for 48 hours, the compressive strength of G1A was increased by 326%, 400%, and 78% at 7, 28 and 56 days respectively. The G2H mortar followed the same trend of G2A, by applying the heat curing, there was an increase of 770%, 312%, and 32.3% at 7-, 28- and 56-days' compressive strength respectively. This result indicates that heat curing had a significant influence on the 7 and 28-day compressive strength of the AAM made by the sodium metasilicate, but the improvement of 56-day strength was lower than the earlier strength.

For G2A and G2H mortars, while applying heat curing, there was a remarkable increase in the 7day compressive strength. Improvement of 583% and 462% was observed in the compressive strength of G2AO and G2HO respectively at the aforementioned age. At 28 days, there was a slight increase but at 56 days, the heat curing had a negative impact on the compressive strength. The compressive strength was dropped from 46 MPa to 39.9 for G2A and from 46.6 MPa to 43.3 MPa for G2H. This result showed that heat curing had a significant influence on the early strength of G2A and G2H mortar but for the later ages the compressive strength of the samples cured at ambient temperature and at incubator had very little difference. This may be due to the higher content of SiO2 which helped the mortar to achieve its late strength without curing at the higher temperature. Therefore, sodium disilicate with a lower concentration than the conventional one

will be effective for making AAM for the sake of improving compressive strength at ambient temperature curing.



Figure 6.13 Effect of the curing condition on the compressive strength

6.5 Conclusion

This research was conducted to develop the one-part mixing method using low calcium fly ashbased AAM at ambient curing conditions. Low-calcium fly ash-based AAM has some drawbacks such as high setting time and low compressive strength when cured at ambient temperature. Replacing some percentage of fly ash with slag can solve the problems but using slag increase the shrinkage of the materials.

Therefore, this research studied the influence of the types and concentration of the sodium silicate solution on the setting time, flowability and compressive strength properties of the AAM. These two types of sodium silicate were sodium metasilicate with a SiO₂:Na₂O ratio of 1 and sodium disilicate with a modulus of 2. Moreover, the effect of the sucrose, temperature of raw materials and curing conditions were also investigated.

From the study the following conclusion can be written;

- 1. SiO₂:Na₂O ratio of the 32% concentrated sodium silicate solution doesn't have a significant influence on the setting and flowability of the paste and mortar samples. The final setting time of the materials was lower than OPC.
- 2. The influence of the SiO₂:Na₂O is prominent on the compressive strength of the AAM. The compressive strength at 28 and 56 days are increased 4 and 2 times respectively when sodium disilicate is used instead of sodium metasilicate.

- 3. The concentration of the sodium silicate has a considerable influence on the setting time of the AAM paste. Using 54% sodium disilicate increase the initial and final setting time to more than 30 hours. But while reducing the concentration, the initial and final setting is reduced to less than 10 hours. Conventionally fly ash-based-AAM made by around 50 concentrated sodium silicate solutions.
- 4. The 54% concentrated or the conventional type AAM has a much lower flow rate than the one which is made by a lower concentration of sodium silicate solution.
- 5. For the ambient temperature curing, a higher concentrated sodium silicate-based AAM has much lower compressive strength at 28 days. But when heat curing is applied, the compressive strength is much higher for the AAM made by a higher concentrated sodium silicate solution.
- 6. Sucrose significantly increased the initial and final setting times of the AAM.
- 7. Using sucrose decreases the flow rate and compressive strength of low calcium fly ashbased fly ash AAM
- 8. The preheating of the fly ash and sand reduces the initial and final setting time of AAM using sodium metasilicate.
- 9. Hot mixing condition breaks the viscosity of the conventional AAM, hence improving the flow rate.
- 10. The temperature of the materials has a significant influence on the compressive strength of the AAM made by the sodium metasilicate but for the sodium disilicate-based AAM, the effect is not so obvious.
- 11. Heat curing has a significant influence on the initial strength of both sodium metasilicate and sodium disilicate-based AAM. But the AAM made by the sodium disilicate solution has higher compressive strength at ambient temperature curing conditions than in heat curing conditions.
- 12. Sodium silicate with SiO₂:Na₂O ratio of 2 (sodium disilicate) with a lower concentration than the conventional one, can set earlier and acquire higher compressive strength at ambient temperature curing conditions. Using this material can be effective both in one-part and two-part mixing methods. Dry sodium disilicate solubility needs to be checked while using it for making one-part AAM.

References

[1] Provis, J.L., Lukey, G.C., van Deventer, J.S.J. (2005). Do geopolymers actually contain nanocrystalline zeolites? —a reexamination of existing results. Chemistry of materials 17(12), 3075–3085.

[2] Ryu, G.S., Lee, Y.B., Koh, T.K., Chung, S.C. (2013). The mechanical properties of fly ashbased geopolymer concrete with alkaline activators. Construction and Building Materials, 47, 409-418. DOI: 10.1016/j.conbuildmat.2013.05.069

[3] Chi, M., Chang, J., Huang, R. (2012). Strength and Drying Shrinkage of Alkali-Activated Slag Paste and Mortar. Advances in Civil Engineering, vol. 2012. https://doi.org/10.1155/2012/579732

[4] Tajunnisa, Y., Sugimoto, M., Sato, T., Shigeishi, M. (2017). A study on factors affecting geopolymerization of low calcium fly ash. International Journal of GEOMATE, 13, 100-107. DOI: 10.21660/2017.36.84153

[5] Tajunnisa, Y., Sugimoto, M., Uchinuno, T., Sato, T., Toda, Y., Hamasaki, A., Yoshinaga, T., Shida, K., Shigeishi, M. (2017). Effect of GGBFS and micro-silica on mechanical properties, shrinkage and microstructure of alkali-activated fly ash mortar. International Journal of GEOMATE, 13, 87-94. DOI: 10.21660/2017.39.11341

[6] Gruskovnjak, A., Lothenbach, B., Holzer, L., Figi, R., Winnefeld, F. (2006). Hydration of alkali-activated slag: Comparison with ordinary Portland cement. Advances in Cement Research. 18. 119-128. DOI: 10.1680/adcr.2006.18.3.119

[7] Han, F., He, X., Zhang, Z., Liua, J. (2017). Hydration heat of slag or fly ash in the composite binder at different temperatures. Thermochimica Acta, 655, 202-210. DOI: 10.1016/j.tca.2017.07.002

[8] Hardjito, D., Wallah, S., Sumajouw, D., Rangan, B. (2004). Factors influencing the compressive strength of fly ash-based geopolymer concrete. Civil Engineering Dimension, 6, 88-93. Available at:

https://www.researchgate.net/publication/26844029_Factors_influencing_the_compressive_stren gth_of_fly_ash-based_geopolymer_concrete

[9] Xie, J., Kayali, O. (2016). Effect of superplasticizer on workability enhancement of Class F and Class C fly ash-based geopolymers. Construction and Building Materials, 122.36-42. DOI: 10.1016/j.conbuildmat.2016.06.067

[10] Hassan, A., Arif, M. & Shariq, M. (2019). Effect of curing condition on the mechanical properties of fly ash-based geopolymer concrete. SN Applied Sciences. 1, 1694. DOI: 10.1007/s42452-019-1774-8

[11] JIS R. (2015). 5201 Physical testing methods for cement.

[12] Provis, J.L. (2009). Activating solution chemistry for geopolymers. In J.L. Provis, J.S.J. van Deventer. (Eds.), Geopolymers: Structures, Processing, Properties and Industrial Applications, Woodhead, Cambridge, UK, pp. 50-71. DOI: 10.1533/9781845696382.1.50

[13] Xie, J., Kayali, O. (2016). Effect of superplasticiser on workability enhancement of Class F and Class C fly ash-based geopolymers', Construction and Building Materials, 122, 36-42. DOI: 10.1016/j.conbuildmat.2016.06.067

[14] Criado, M., Palomo, A., Fernández-Jiménez, A., Banfill, P.F.G. (2009). Alkali activated fly ash: Effect of admixtures on paste rheology', Rheologica Acta, 48, 447-455. DOI: 10.1007/s00397-008-0345-5

Chapter 7

Fundamental Investigation of Recycled Concrete using Paper Sludge Ash and Silica Fume

7.1. Introduction

Portland cement concrete is the most used construction material in the world. Annually, a huge amount of waste concrete is generated due to the demolition and reconstruction process. It is estimated that 900 million tons of waste concrete are generated in the USA, India, Europe, and Japan. Most of this waste is trashed and disposed of in landfills which creates environmental problems [1].

Recycled Concrete Aggregate (RCA) is a concrete material that is produced by modification of waste concrete, this material is not widely used as aggregate due to its higher water absorption and lowers specific gravity, consequently, strength and durability are negatively affected. Different kinds of substances such as silica fume, fly ash, and steel fiber are used as an admixture or as partial cement substituents to enhance the strength and durability of recycled aggregate concrete [2], [3] [4].

Waste paper sludge is another material that can be recycled like other pozzolans such as ash fly ash [5], after incinerating paper sludge at approximately 800 °C, ash will be produced that have reactive silica and can chemically react will cement and can improve some properties of concrete [6].

Silica fume (SF) is an ultrafine by-product of silicon processing industries. It has an amorphous shape and highly reactive silica. Silica fume has a surface area of 15-25 m²/gram. This by-product increases the Si/Ca ratio in the binder which improves the permeability of the paste and hence, increases the strength and durability of concrete [7]. This research will further study the by-product by conducting an Acoustic emission (AE) behavior test on concrete samples containing 5 and 10% silica fume.

Each of the binder (SF and PSA) need to be used together with RCA to study their influence on some properties of recycled aggregate concrete. Moreover, each of the binders and recycled aggregate also need to be used separately to understand their single effect on the properties of Portland cement concrete.

This experiment aims to study the influence of silica fume and paper sludge ash on some of the properties such as flowability and compressive strength of recycled aggregate concrete. Moreover, the acoustic emission behavior of these kinds of concrete will also be investigated as there isn't any past experiment of studying the AE behavior of concrete or recycled concrete using paper sludge ash and silica fume.

7.1.2 Significance of the research

- Enhance the tendency of people to utilize paper waste and concrete waste effectively to reduce disposal amount for landfills.
- Energy and heat can be produced while burning paper sludge, and by secondary ignition and incineration at a temperature of about 800, it can be converted to pozzolan, which can be partially replaced with cement.
- Consequently, the rectification and exploitation of both wastes can have economic and environmental advantages.

7.2 Objectives

The following properties of concrete with different percentages of paper sludge ash, silica fume, and recycled aggregate are investigated and then compared with the conventional or reference concrete.

- Workability
- Compressive strength
- Acoustic Emission behavior

7.3 Experiment

7.3.1 Materials

The materials which were used for the making of the samples consisted of the binders, aggregates, water, and admixture. The specifications of the materials are explained as follow:

7.3.1.1 Binders

Ordinary Portland cement (OPC), Paper sludge ash, and silica fume with specific gravities of 3.15, 2.43, and 2.22 respectively, were used as binders for making the samples.

Figures 1 and 2 show the SEM (scanning electron microscope) of OPC and PSA. Both the materials have an amorphous molecular structure.



Figure 7.1 SEM of OPC and PSA

XRF (X-ray fluorescence) was conducted on each of the binders to find chemical compositions and the result is shown in Table 1.

Oxides	OPC	PSA	Silica Fume
	(%)	(%)	(%)
Na ₂ O	0.217	0.372	0.213
MgO	0.964	2.968	0.390
Al ₂ O ₃	4.234	12.282	0.237
SiO ₂	17.899	16.906	96.536
P ₂ O ₅	0.174	1.162	0.082
SO ₃	3.009	2.412	0.223
K ₂ O	0.517	0.314	0.837
CaO	69.045	61.068	0.222
TiO ₂	0.279	0.615	1.079
MnO	0.076	n/a	n/a
Fe ₂ O ₃	3.411	1.611	0.100
CuO	0.041	0.063	0.017
ZnO	0.070	0.068	n/a
SrO	0.048	0.039	n/a
ZrO ₂	0.017	0.016	n/a

Table 7.1 XRF of the cement, paper sludge ash, and silica fume

7.3.1.2 Aggregates

Aggregate is an important component of concrete. It has a major role in the specification of the properties and quality of concrete. In this experiment, around 80% of the mass of the concrete was occupied by the aggregate. Both fine and coarse aggregates are described as below;

a) Fine aggregate

Sea sand was used as fine aggregate with oven-dry density and water absorption of 2.5 and 2.87 respectively

Table 7.2 Size	e distribution	of fine aggregate
----------------	----------------	-------------------

Size of sieve openings (mm)	Passing (%)
9.5	100
4.75	100
2.36	98.42
1.18	81.82
0.6	52.26
0.3	24.74
0.15	7.1

b) Coarse aggregate

Crushed stones and recycled aggregates were used as coarse aggregate. Specific gravity and water absorption are shown in Table 3 below.

Physical property	Crushed stone	Recycled aggregate
Oven-dry density (g/cm ³)	3.02	2.76
Water absorption	0.56	2.4

Table 7.3 Properties of coarse aggregate

The pulsed power apparatus was used to produce recycled aggregate. A high grade of recycled aggregate can be discharged when this apparatus is used [8] (Shigeishi, et al., 2013). The recycled coarse aggregate was obtained by applying energy of E/N=6.4 kJ per pulse at a capacitance of 0.8 μ F and the applied voltage of 400 kV satisfied the oven-dry density criterion and the water absorption coefficient criterion when using applied energy of 384 kJ. Thus, the frequency of pulsed discharge was 60 times underwater. Consequently, the high-grade recycled aggregate was produced which satisfied JIS regulation class H (A 5021) for high-grade recycled concrete aggregate, as this regulation requires oven-dry density for recycled aggregate to be at least 2.5 gr/cm³ and maximum water of absorption of 3%.



Figure 7.2 Recycled aggregate

Size of sieve opening	Passing (%)				
(mm)	Crushed stone	Recycled aggregate			
25	100	100			
20	96.31	100			
15	84.07	94.66			
10	27.74	56.71			
5	0.71	0.42			
2.5	0.04	0.05			
1.2	0.035	0.03			

Table 7.4 Size distribution of coarse aggregate

7.3.2 Procedure and experiment

The procedure of the sample preparation and conduction of the experiment are described in the following subsections.

7.3.2.1 Mix proportion

The w/c ratio was maintained to 0.55, while sand/ aggregate was kept to 0.48. 15 types of samples were prepared. The mix proportion is described in Table 5. To study the effect of paper sludge ash on the properties of concrete, it was replaced with 5 and 10 percent of cement, and the same amount of cement was replaced by another pozzolan which is silica fume. Silica fume is a widely used material in the world for the enhancement of some characteristics of concrete. To investigate the influence of recycled aggregate on the properties of concrete samples, they replaced 10 and 20 percent of natural aggregates. Polycarboxylate superplasticizer was used as a water-reducing agent, and to increase the flowability of concrete, the amount of usage was kept the same (1 percent of the weight of cement) for all sample types.

No.	Sample name	Cement	PSA	Silica fume	Fine aggregate	Coarse aggregate	RCA	Water
1	CON	318	0	0	841	1070.00	0	175.00
2	CA5	300.50	15.82	0	841	1070.00	0	173.97
3	CS5	299.90	0	15.78	841	1070.00	0	173.63
4	CA10	283.20	31.47	0	841	1070.00	0	173.07
5	CS10	281.90	0	31.32	841	1070.00	0	172.27
6	10RC	318	0	0	841	956.00	106.00	175.00
7	10RA5	300.50	15.82	0	841	956.00	106.00	173.97
8	10RS5	299.90	0	15.78	841	956.00	106.00	173.63
9	10RA10	283.20	31.47	0	841	956.00	106.00	173.07
10	10RS10	281.90	0	31.32	841	956.00	106.00	172.27
11	20RC	318	0	0	841	842.00	212.00	175.00
12	20RA5	300.50	15.82	0	841	842.00	212.00	173.97
13	20RS5	299.90	0	15.78	841	842.00	212.00	173.63
14	20RA10	283.20	31.47	0	841	842.00	212.00	173.07
15	20RS10	281.90	0	31.32	841	842.00	212.00	172.27

Table 7.5 Mix proportion (kg/m³)

7.3.2.2 Mixing

A concrete mixer was used to mix the materials. At first OPC, the fine and coarse aggregate was blended for 1 minute and after adding water and the superplasticizer, they were mixed for 1.5 minutes. But in the case of using PSA and SF, cement was first blended with sand, then for effective mixing, the cement substituents were poured in different points and blended for 1 minute. Afterward, coarse aggregate and then water was added to the mixer for mixing.



Figure 7.3 Slump test and samples after casting in molds

7.3.2.3 Slump test, specimen casting, and curing

The slump test was conducted according to JIS A 1101. The test was done to study the workability of fresh concrete. A cone with a height of 30 cm, with a bottom diameter of 20 cm and an upper 10 cm was used. For each slump test, the cone was filled by mortar in 3 layers and each layer was tempted 25 times by a 16 mm rod. Afterward, the cone was carefully removed and the height difference is measured as a slump. All the samples were cylindrical, with a dimension of 100 mm diameter and 200 mm height. The molds were removed after 24 hours and samples were immersed in water for curing at 18-22 °C temperature until the day of their testing.

7.3.2.4 Compression test

The compressive test was carried out on hardened concrete samples to find compression strength. Loads were applied on each of the samples until their failure, and the peak loads were then recorded. This test was performed on all 150 cylindrical samples (100 mm*200 mm) after 28 and 56 days of curing.

7.3.2.5 Acoustic Emission (AE) behavior test

An AE sensor was attached to each of the samples during the compressive strength test. Due to the compression, stresses occurred inside the samples. The micro-cracks were produced by the stresses emitted by sound waves with different intensities. Above 40 decibels acoustics or elastic wave were considered to be produced by those cracks are called acoustic emission (and abbreviated as AE). AE waves propagate through concrete and can be detected on a surface by an AE sensor, which turns the

vibrations into an electrical signal. The propagation of fracture sound was originally referred to as AE since it is acoustic and audible [9] (Ohtsu, 1995).



Figure 7.4 Parameters of acoustic emission waveform [10]

7.4 Result and discussion

7.4.1 Workability

Figure 7.6 shows the slump value of each sample. The workability of concrete was increased with the 5% silica fume but slightly decreased with the 5% PSA. Both the 10% PSA and 10% silica fume significantly decreased the slump of concrete. The 10 and 20% RCA improved the workability of the samples. The 5% silica fume didn't decrease the flowability of the 10 and 20% recycled concrete but the workability of the samples was notably decreased with PSA and 10% silica fume concrete.



Chapter 7. Fundamental Investigation of Recycled Concrete using Paper Sludge Ash and Silica Fume

Figure 7.5 Slump value of the concrete samples



7.4.2 Compressive strength

Figure 7.6 The 28-day compressive strength



Figure 7.7 The 56-day compressive strength

Figures 7.7 and 7.8 show the compressive strength of the samples at 28 and 56 days. Maximum, minimum, and average strengths are shown on the charts and an average value is accepted as compressive strength for each type of sample.

Both the silica fume and the PSA increased the 28 and 56-day compressive strength. It can be said that pozzolanic reaction between ingredients of the PSA, SF, and the OPC. From the figures, it can be understood that there wasn't any significant decrease in compressive strength when recycled aggregate was used, but on the other hand, the 20% RCA increased the 56-day strength. This may be due to previously attached mortar on the surface of the RCA which helped to strengthen the bond between the cement and the RCA. Both the SF and the PSA improved the strength of the RCA concrete, especially, when they were used with the 20% recycled aggregated concrete.

7.4.3 Acoustic Emission (AE) behavior

Hits vs stress diagrams of the concrete samples were drawn to study the robustness of the samples. Hits which are the accumulation of acoustic data that exceed the threshold were achieved throughout the loading process. Generally, sound concrete has low numbers of AE hits at the initial stage of stress production and the amount exceeds at the final stage. Figure 9 shows the relation between AE hits and stress levels during compression loading for all types of concretes. There is no significant difference in AE behavior between conventional concrete, silica fume concrete, and both the 10% and 20% RCA concrete. AE hits were found to be consistently low at the beginning of loading and were higher when the stress reached around 100%. Moreover, AE activity and number of cumulative hits were lower in the PSA concrete, this indicates the robustness of this kind of concrete.


Chapter 7. Fundamental Investigation of Recycled Concrete using Paper Sludge Ash and Silica Fume

Figure 7.8 Relationship between AE hits and stress level for the concrete samples

Chapter 7. Fundamental Investigation of Recycled Concrete using Paper Sludge Ash and Silica Fume

7.5 Conclusion

According to literature, recycled aggregate has a higher porosity and lower density has a negative influence on some properties such as strength and durability of recycled aggregate concrete. In this experiment, paper sludge ash and silica fume were used to modify the properties of recycled concrete, moreover, the effect of both pozzolans are also investigated on OPC concrete. The following conclusion was acquired from the study;

- 1. The workability of concrete is decreased with an increase in the amount of silica fume and paper sludge ash.
- 2. The recycled aggregate produced by pulsed power improves the workability of fresh concrete.
- 3. Both PSA and SF improve the compressive strength of conventional and recycled aggregate concrete. This indicates that both of the materials have shown their pozzolanic activity inside the mixture.
- 4. Both the 10 and 20% recycled aggregate concrete improve the compressive strength of the samples. And this is maybe because of the surface roughness of the RCA due to the previously attached mortars.
- 5. The acoustic emission activities of all types of samples were lower at the early stage of loading. The cumulative hits in PSA concrete were lower and this shows that PSA concretes were denser and had higher robustness than the other type of concrete samples.
- 6. From this research, it can be concluded that both silica fume and paper sludge ash can improve the properties of concrete. although, the workability is negatively affected by a higher percentage of the substituents, therefore, there is a need for studying the influence of other types of admixtures on the workability of such kinds of materials.
- 7. High-quality recycled aggregate didn't have a negative influence on the properties of concrete but can improve some of the properties of the concrete.
- 8. The effect of the pozzolans on properties of concrete made by a lower or medium quality of recycled aggregate is also essential to be evaluated.

Future research

In this research, the workability of the samples decreased when the percentage of the PSA and silica fume increased to 10%, therefore, other types of admixtures need to be used and to be studied to overcome this drawback. Moreover, other properties such as durability properties also need to be investigated while using these kinds of cement and aggregate substituents.

There are some kinds of by-products such as rice husk which shows pozzolanic properties when they are burnt at high temperature. These kinds of materials are also better to be studied comprehensively. Chapter 7. Fundamental Investigation of Recycled Concrete using Paper Sludge Ash and Silica Fume

References

[1] Yadhu, G., Devi, S.A. (2015). An Innovative, Study on Reuse of Demolished Concrete Waste. Journal of Civil & Environmental Engineering, 05. DOI: 10.4172/2165-784X.1000185

[2] Buck, A. D. (1977). Recycled Concrete as a Source of Aggregate. ACI Journal Proceedings, 74, 212-219. DOI:10.14359/11004

[3] Yang, J., Du, Q., Bao, Y. (2011). Concrete with recycled concrete aggregate and crushed clay bricks. Construction and Building Materials, 25(4), 1935-1945. DOI: 10.1016/j.conbuildmat.2010.11.063

[4] Çakır, Ö., Sofyanlı, Ö. Ö. (2015). Influence of silica fume on mechanical and physical properties of recycled aggregate concrete. HBRC Journal,11(2), 157-166. DOI: 10.1016/j.hbrcj.2014.06.002

[5] Luga, E., Paja, A., Atis, C. D. (2017). An Investigation on the Partial Replacement of Portland Cement with Kosovo Fly Ash in Cement Mortars. MATTER: International Journal of Science and Technology, *3*(3), 125-141. DOI: 10.20319/mijst.2017.33.125141

[6] Sajad, A. (2013). Study of Concrete Involving Use of Waste Paper Sludge Ash as Partial Replacement of Cement. IOSR Journal of Engineering, 3(11), 06-15. DOI: 10.9790/3021-031130615

[7] Chakravarthy H.G, N., Ali, F., I. (2015). Effect of Silica Fume on Partial Replacement of Cement on High Strength Concrete. MATTER: International Journal of Science and Technology, 1(1), 275-284. DOI: 10.20319/mijst. 2016.s11.275284

[8] Shigeishi, M., Namihira, T., Iizasa, S., Ishimatsu, K., Arifi, E., Amoussou, R. (2013). Pulsed Power Application to Production of Recycled Aggregate. In: Proceedings of the 3rd Sustainable Construction Materials and Technologies. Japan Concrete Institute.

[9] Ohtsu, M. (1995). The history and development of acoustic emission in concrete engineering. Magazine of concrete research 48: 321-330.

[10] Wikimedia Commons, the free media repository. Retrieved 02:09, March10,2021. from <u>https://commons.wikimedia.org/w/index.php?title=File:Figure5.jpg&oldid=4649698</u> <u>26</u>

Chapter 8

Conclusion

Global warming is a major issue nowadays. Ordinary Portland cement (OPC) production is responsible for the emission of 8% CO2 of the world. Moreover, some by-products such as fly ash are disposed to landfills which create an environmental problem. Using alkali-activated material (AAM) as an alternative for OPC can have both environmental and economic advantages. Moreover, they have some extraordinary properties such as high strength, high resistance to fire, heat, and acid, and low shrinkage. With having all the aforementioned properties, their usage is still limited. Difficulties in handling corrosive alkaline solution and high viscosity of sodium silicate and complicated technology are the problems that make the material to be made in factories as a precast element. This causes huge costs and a massive reduction in the production of AAM.

Efforts have been made to make ease in the production of AAM, therefore, a one-part mixing method has been studied by some researchers in which solid alkaline activators are premixed with aluminosilicate precursors and then water is added to them for making the binder. This method is the same as making an OPC binder. But some studies concluded that the AAM made by the one-part mixing method has lower quality than the one made by the conventional or two part-mixing method. Therefore, this research aims to find the root of the problem and find a solution for that. Furthermore, this study aims to develop the properties of low-calcium fly ash-based-AAM cured at ambient temperature conditions.

In chapter 3, the properties of AAM made by the one-part mixing method were characterized. The influence of the mixed proportion of aluminosilicate precursors such as ash fly ash, slag, and silica fume, the molarity of sodium hydroxide, AA ratio was studied. Furthermore, the properties of AAM made by the one-part mixing method were compared with the OPC mortar and AAM made by the conventional method in which sodium silicate solution was used with a higher concentration. Although compressive strength of the AAM made by the one-part mixing method increased with the increment of the slag in composition their quality was lower than the OPC and conventional type AAM. Moreover, there was a quick setting occurred with the one-part mixed AAM containing a high percentage of slag. This cause workability problem in casting the material into the molds.

Chapter 4 focuses on the retarding influence of sucrose on the AAM. As it was mentioned in the previous paragraph, the workability problem that occurred with the AAM contained a high amount of slag. This chapter revealed that using sucrose retards the setting time of the AAM, hence the flowability and compressive strength of the material was improved. A better spread of the aluminosilicate and alkaline activators through the mixture due to the better flowability of the fresh mortar was mentioned as a reason for the improvement of strength by sucrose.

In chapter 5, the flowability and compressive strength comparison of AAM made by one-part and two-part mixing methods were discussed. Both kinds of materials had the same mix proportion and composition. The result revealed that AAM made by the one-part mixing method had much lower flowability and compressive strength than the one made by the two-part mixing method. In the literature review, it was found that undissolved particles of sodium metasilicate were observed while studying the microstructure of one-part mixed AAM.

Therefore, the solubility of solid alkaline activators such as sodium hydroxide pellets and granular sodium metasilicate was studied. The duration of mixing was set to 5 minutes. The result showed that sodium hydroxide pellets were quickly dissolved at 25 °C but the sodium silicate couldn't be dissolved in the solvent at the aforementioned temperature. Then the temperature of the solvent was increased and improved the solubility of the material was observed. The total dissolution occurred when the temperature of the solvent was reached 40°C. So it was concluded that the temperature of a solvent needs to be at around 40°C while the one-part mixing method is applied. Remember that the aforementioned temperature is adopted when 32% sodium metasilicate is used. For the lower and higher concentration of sodium silicate, the temperature can be changed.

Moreover, in this chapter, a solution made by granular sodium metasilicate was characterized. It was found that sodium metasilicate was fluent and without viscosity. But it was susceptible to a temperature around 25°C. Because crystallization could occur inside the solution. But after putting it inside an incubator at a higher temperature the crystallization could disappear. The fresh mortar of AAM made by sodium metasilicate was much influenced by the ambient temperature and quick hardening or setting had a negative influence on the flowability of the mortar. Therefore, it was decided to use the hot mixing method. In the hot mixing method, all the raw materials were used at a temperature of around 40°C. Improvement in workability was observed after adopting the method.

Low-calcium fly ash-based-AAMs have some extraordinary geopolymeric properties such as high fire resistance and low shrinkage. But it is conventionally made by curing at a higher temperature inside an oven and also sodium silicate with a concentration of around 50% is used in the material. But they have a very high setting time and low compressive strength when cured at ambient temperature. In chapter 6, the influence of sodium silicate types and concentration on the setting time, flowability, and compressive strength properties of the AAM was discussed. The concentration of 32% and 56% were investigated. In addition, the influence of the raw material temperature, sucrose, and curing condition was studied on the aforementioned properties of low-calcium fly ash-based AAM. This study aimed to solve the high setting time and low compressive strength problem of low-calcium fly ash-based AAM cured at ambient temperature conditions.

The concentration of sodium silicate had a significant influence on the setting time, flowability and compressive strength of the AAM cured at ambient temperature conditions. 32% sodium disilicate considerably improved all the aforementioned properties of the AAM. This was maybe due to the better reactivity of fly ash in an alkaline environment with a decreased PH value. There is a need for some further study to find the reason related to the phenomenon. In this study, sucrose didn't improve the properties of the AAM. This was contrary to the previous study in which

sucrose improved the flowability, compressive strength, and drying shrinkage of AAM containing a high amount of slag. But in this experiment, as the reactivity of low-calcium fly ash is very low, and the flowability of the AAM was good enough for the dispersing of the main constituents, the use of sucrose was not required.

As it was found that sodium disilicate with a lower concentration than 50% can be effective for decreasing the setting time and improvement of flowability and compressive strength of low-calcium fly ash-based AAM cured at ambient condition, the solubility of this type of sodium silicate is also required to be studied before using its solid form for the making of one-part mixed AAM.

Chapter 7 investigated the influence of recycled concrete aggregate produced by pulse power technology, Silica fume, and paper sludge ash on the workability, compressive strength, and acoustic emission behavior of concrete. In the experiment, 5 and 10% of cement were replaced by PSA and silica fume, and 10 and 20% of coarse aggregate were replaced by recycled aggregate. Both the PSA and SF showed their pozzolanic activity as both the OPC substituents improved the Strength and acoustic emission behaviors of concrete. Moreover, the high-quality recycled aggregate also modified the properties of concrete.